

ESEARCH REPORT

HOMESAVE: BUILDING INVESTMENTS

IN HOUSING ASSETS: BACKGROUND REPORT





HOME TO CANADIANS
Canadä

CMHC—HOME TO CANADIANS

Canada Mortgage and Housing Corporation (CMHC) is the Government of Canada's national housing agency. We help Canadians gain access to a wide choice of quality, affordable homes.

Our mortgage loan insurance program has helped many Canadians realize their dream of owning a home. We provide financial assistance to help Canadians most in need to gain access to safe, affordable housing. Through our research, we encourage innovation in housing design and technology, community planning, housing choice and finance. We also work in partnership with industry and other Team Canada members to sell Canadian products and expertise in foreign markets, thereby creating jobs for Canadians here at home.

We offer a wide variety of information products to consumers and the housing industry to help them make informed purchasing and business decisions. With Canada's most comprehensive selection of information about housing and homes, we are Canada's largest publisher of housing information.

In everything that we do, we are helping to improve the quality of life for Canadians in communities across this country. We are helping Canadians live in safe, secure homes. CMHC is home to Canadians.

You can also reach us by phone at 1 800 668-2642 (outside Canada call (613) 748-2003)
By fax at 1 800 245-9274 (outside Canada (613) 748-2016)

To reach us online, visit our home page at www.cmhc.ca

Canada Mortgage and Housing Corporation supports the Government of Canada policy on access to information for people with disabilities. If you wish to obtain this publication in alternative formats, call I 800 668-2642.

BACKGROUND REPORT

Study of Individual Development Accounts for Housing - A National Demonstration Project -



Study Prepared for: Canada Mortgage and Housing Corporation and the National Secretariat on Homelessness

By:



February 2003

PREFACE

This project was funded by Canada Mortgage and Housing Corporation (CMHC) and Human Resources Development Canada (HRDC). The contents, views and editorial quality of this report are the responsibility of the author and CMHC and HRDC accept no responsibility for the contents, views and editorial quality or any consequences arising from the reader's use of the information, materials and techniques described herein.

The copyright remains with CMHC, HRDC and with the contractor, Social & Enterprise Development Innovations (SEDI).

Cette recherché a été finance par la Société Canadienne d'hypothec et de logement (SCHL) et Developpement des ressources humaines Canada (DRHC). La qualité rédactionnelle de ce rapport, son contenu et les opinions qui y sont émises sont de l'auteur. La SCHL et DRHC ne saureaient en assumer la responsabilité pour la qualité redactionnelle, le contenue et les opinions et se dégagent de toute obligation rélativement aux conséquences resultant de l'utilisation que le lecteur pourrait faire des renseignements, des matériaux ou des techniques qui y sont décrits.

SEDI is thankful to all of those in the asset-building and related fields who provided valuable insights, professional contributions and welcomed encouragement for this project.

TABLE OF CONTENTS

1. INTRODUCTION	1
1.1. OUTLINE OF REPORT	1
2. IDA PROGRAMS	3
2.1. Existing Demonstration Projects in the US and Canada	
2.1.1 Existing Demonstration Projects in the US and Canada 2.1.1. American Dream Demonstration	
a) Summary of Results	
Participation	
Savings	
<u>Costs</u>	
b) Other Findings	
2.2.2.Assets for Independence Demonstration.	
2.2.3. learn\$ave	
a) Parameters	
b) Results	
2.2. OTHER US IDA INITIATIVES	9
2.2.1. Savings for Working Families Act	9
2.2.2. State-Sponsored Programs	10
a) State Supports	10
b) Local Programs	10
Organizations	11
Participants	11
Income Eligibility	
Authorized uses	
Matching Provisions	
2.3. OTHER RELATED US PROPOSALS.	
2.4. Canadian Programs	
2.4.1. Calgary's Fair Gains & Owen Hart Homeowners Program	
a) Fair Gains	
b) Owen Hart Homeowners Program	13
Observations	
2.4.2. Winnipeg's Housing IDA Program	
a) Project Conditions	
b) Participation	
2.5. LOCAL US PROGRAMS	
2.5.1. Methodology	
2.5.2. Organizations	
a) Background in Affordable Housing	
b) Consortia	
c) Experience with Education and Counselling	
d) Permanence	
2.5.3. Uses for Savings 2.5.4. Details of Programs	
a) Savings Rates	
b) Match Rates	
c) Fligibility Requirements	∠∪

d) Admission Process	21
2.5.5. Sources of Funding	22
2.5.6. Use of Housing Assistance	23
2.5.7. Involvement of Banks	24
2.5.8. Monitoring.	25
2.5.9. Impact on Mortgage Defaults.	
2.5.10. Special Programs	
S. CANADIAN GOVERNMENT HOMEOWNERSHIP PROGRAMS	27
3.1. Federal Programs	
3.1.1. RRSP Home Buyers' Plan	
3.1.2. CMHC's 5% Down Program	
3.2. Provincial Programs.	
3.2.1. Ontario Home Ownership Savings Plan	
3.2.2. New Brunswick's Home Ownership Program	
3.2.3. Saskatchewan's Neighbourhood Home Ownership Plan	
3.3. MUNICIPAL PROGRAMS	
3.3.1. Surrey's Home Ownership Assistance Program	
3.3.2. Winnipeg's Rent-to-Own Program.	
3.3.3. Winnipeg's Housing Opportunity Partnership	
3.3.4. Montreal's Domi-cible Program	
3.4. Affordable Housing Agreements	35
STAKEHOLDERS' MEETINGS	37
4.1. HOMEOWNERSHIP ASSISTANCE 4.1.1. Target Population	
4.1.2. Eligibility Criteria	
4.1.3. Subsidy Protection.	
4.1.4. Public and Political Support.	
11	
4.1.5. Education and Counselling Programs 4.1.6. Other Concerns	
4.2. Other Housing Options	
4.2.1. Home Repairs	
4.2.2. Equity Co-ops.	
4.2.3. New Construction	
4.2.4. Rental Housing	
4.2.5. Non-Profit Co-ops	
4.2.6. Secondary Apartments.	
4.3. LOCAL PARTICIPATION.	49
S. RENT BANKS	=4
5.1. TORONTO: RENT BANK (PILOT PROGRAM)	51
<u>5.1.1. Results</u>	51
5.1.1. Results	51 52 53
5.1.1. Results 5.1.2. Recommendations 5.2. TORONTO: RENT BANK ("EXPANDED" PROGRAM)	51 52 53
5.1.1. Results	51 52 53 53
5.1.1. Results. 5.1.2. Recommendations. 5.2. TORONTO: RENT BANK ("EXPANDED" PROGRAM) 5.2.1. Conditions 5.2.2. Results	51 52 53 53 54
5.1.1. Results	515253535455

5.4.1. Results	58
5.4.2. Other Services	58
5.5. Hamilton: Housing Emergency Loan Program (HELP)	59
<u>5.5.1. Results</u>	60
5.5.2. Observations	60
5.6. CALGARY: SECURITY DEPOSIT PROGRAMS	61
6. HOMEBUYER EDUCATION AND COUNSELLING PROGRAMS	61
6.1. Background	62
6.2. Experience in the US	63
<u>6.2.1. Stakeholders</u> .	63
6.2.2. Clientele	64
6.2.3. Services	
6.2.4. Involvement as "Intermediaries".	65
6.2.5. Role of Lenders	
<u>6.2.6. Tele-HEC</u>	
<u>6.2.7. Costs</u>	
6.2.8. Effectiveness	
6.3. Experience in Canada	67
7. ASSESSMENT OF LICOS	69
7.1. Background	69
7.2. LICOS AS MEASURES OF POVERTY	70
<u>7.2.1. Merits</u>	71
7.2.2. Limitations	72
7.2.3. Relative vs. Absolute	72
7.2.4. Alternatives	
a) Low Income Measure (LIM)	
b) Market Basket Measure (MBM)	
c) Basic Needs Measure	
7.3. LICOS AS MEASURES FOR HOMEOWNERSHIP	
7.3.1. Comparison of the LICOs to Mortgage Requirements	76
APPENDIX 1: LIST OF PARTICIPANTS AT THE STAKEHOLDERS' N	<u>MEETINGS</u>
	76
APPENDIX 2: REFERENCES	88

1. Introduction

The report contains the background information collected and reviewed as part of the field work undertaken to identify the possible resonance and appropriateness of a potential national demonstration project aimed at testing "Individual Development Accounts" (IDAs) to provide lower-income households with greater access to affordable housing. The project itself has been called *Home*\$ave.¹

This study was sponsored by Social and Enterprise Development Innovations (SEDI) using funding provided by Canada Mortgage and Housing Corporation and the National Secretariat on Homelessness. SEDI is a non-profit organization engaged in exploring social policy innovations, particularly in the asset-building field.

The major conclusions and findings for this study are presented in the accompanying Main Report: Study of Individual Accounts for Housing, A National Demonstration Project – Home\$ave-February 2003.

The research on the consumer market response to the *Home*\$ave concept is contained in a separate document: *Home*\$ave *Participant Consultative Research (Final Version)*, April 2002.

1.1. Outline of Report

This report contains documentation that was reviewed and thought pertinent to include as supporting information to that contained within the aforementioned *Main Report: Study of Individual Accounts for Housing, A National Demonstration Project – Home*\$ave- February 2003. This report can be read in conjunction with the main findings if additional information or detail is sought.

The documentation within this background report is presented in 7 sections and commences with a review of existing IDA programs in both Canada and the United States. Section 3 provides additional information on Canadian Government homeownership programs that adds support to the premise that housing is seen as an appropriate means to accumulate assets. Section 4 provides details of the Stakeholders meetings that took place in March and April of 2002 and the results of the discussions undertaken on the items related to IDAs for greater access to affordable housing. Section 5 identifies details of six rent bank programs and the effect of the administration and related disbursement of the contribution tot eh individual/families involved. Section 6 provides details the background and experience of Homebuyer Education and councelling programs both in the US and Canada and Section 7 ends by assessing the various poverty measurements utilized today in Canada.

The report concludes by providing a list of participants who attended, and provided input at the Stakeholder meetings held in Halifax, Montreal, Toronto, Winnipeg and Saskatoon, as well as a

¹ Home\$ave is the project name trademarked by SEDI in October 2002.

list of references used to compile information for this study. Additional information relating to the various aspects of IDAs for housing can also be gained from these sources.

2. IDA PROGRAMS

2.1. Existing Demonstration Projects in the US and Canada

The following provides details of existing IDA demonstration programs in the US and Canada.

IDA programs in the US are supported by two on-going demonstration projects namely the *American Dream Demonstration* (ADD) initiated in 1997 by the Corporation for Enterprise Development (CFED) and the *Assets for Independence Demonstration* (AFI) initiated by the federal government in 1999.

In addition, since 1993, 36 states and the District of Columbia have started programs to support IDAs, and 32 states have included them as part of their welfare reform plans.

No survey has been undertaken of the local IDA programs. There are 392 programs registered in CFED's network. According to staff, virtually all of them are involved in asset development for education, micro-enterprise as well as homeownership; but very few have extensive experience in or are exclusively devoted to housing. Many of these are housing organizations that have included IDA as one of the ways available to address affordable housing. Most are social service or business development organizations that have adopted IDAs.²

learn\$ave is a Canadian nation-wide demonstration project which commenced in 2001 and is examining the use of IDAs for adult education. The project, originally initiated, designed and currently administered by SEDI, is being supported through \$35 million provided by the Applied Research Branch of Human Resources Development Canada (HRDC). The local programs are being administered by non-profit organizations in ten communities namely Calgary, Toronto, Winnipeg, Montreal, Vancouver, Halifax, Fredericton, Kitchener-Waterloo, County of Grey Bruce, and Digby and Annapolis County.

2.1.1. American Dream Demonstration

The American Dream Demonstration (ADD) is the first large-scale test of IDAs. The Corporation for Enterprise Development (CFED) was responsible for initiating and coordinating the project [Schreiner, 2001]. The Center for Social Development (CSD) at Washington University in St Louis is responsible for its evaluation.

CFED is a non-profit organization based in Washington DC that promotes asset-building approaches, primarily in low-income and distressed communities, as a strategy for creating economic opportunities.

² Mills, Gregory et al: Evaluation of Asset Accumulation Initiatives, Final Report; Abt Associates: prepared for the US Dept of Agriculture: 29 February 2000. Available at:

http://www.fns.usda.gov/oane/MENU/Published/FSP/FILES/Program%20Design/AssetAcc.pdf

CFED raised approximately \$18 million, for the project from 11 private foundations. Some of the local organizations also have subsequently secured additional funding from the federal program and/or various local sources, including state programs that support IDAs.

Out of this funding, CFED matches the participant savings in the IDAs on a 1:1 basis up to \$500 per account per year. It also provides annual operating grants of \$25,000 to all of the participating organizations, except for the largest operation, Community Action Project of Tulsa County (CAP TC) in Oklahoma, which received \$100,000.

In coordinating this project, CFED also conducted a competitive selection of the participating local organizations, provided technical assistance, undertook policy development and research, hosted annual conferences, opened a web-based network, and published numerous reports.

In ADD there are 14 IDA programs that have been designed, implemented and operated by 13 local organizations across the US. The organizations are located in a range of urban, small cities, small towns and rural areas. CAPTC is responsible for about 1/4 of the participants in this project.

The project was opened to new enrollments in July 1997, and these were allowed to continue until the end of 1999. The project will run until the end of July 2002, after which CFED will no longer match any new deposits (There is one exception to this: CAP TC will run for one more year because the recruitment of participants in this large program took a year longer than expected.) The on-going evaluation of the project is expected to continue for two years after that until at least mid-2004.

After the end of July 2002 (July 2003 for CAP TC), the accounts must be closed or taken over by the local organizations. Those local organizations who have raised federal and local funding will most probably will continue under their auspices. Where that is not possible, and the participants have not utilized the matching funds, they will be permitted to roll over their savings and the matching funds to an Individual Retirement Accounts (IRAs).

These programs share the following main features:

- In all of these programs, the matches are available for withdrawals from the IDAs when used for home purchase, post-secondary education and micro-enterprise development. Some of the programs also match deposits when used for home repair, retirement and job training.
- The match rates in these programs range from 1 to 7 dollars for every deposited dollar withdrawn for an approved use. The average match was just under 2:1.
- Eligibility is limited to households earning no more 200% of the poverty line. This criterion is considered explicitly to target the "working poor" rather than the poor.
- All of the participants are required to receive some financial education. The time provided varied, but the average was about 10 hours.

a) Summary of Results

The following is a summary of the final evaluation report (Schreiner *et al*, 2002) prepared by the Centre for Social Development (CSD), St.Louis, MO. It is based on data collected from 1997-2001, although participants accessed their matched withdrawals until June 30, 2002.

Participation

Nearly 2,400 participants had opened IDA accounts in 14 IDA programs across the United States. Approximately 56% of participants are considered *savers*, meaning they saved more than \$100 as of Dec. 31/2001. The average length of participation was 24.5 months.

These participants are not representative of the low-income population as a whole in the US. In comparison, they contain a higher portion of females (80%), African-Americans (47%), unmarried people (75%), and those with a secondary or higher education (75%), a regular full-time job (58%), and a bank account (79%). These differences likely reflect the explicit targeting to the "working poor" by these programs, and also the populations served by the various local organizations running the programs.

On average, the participants in ADD earned 116% of the poverty line. About 12% were over 200% of the poverty line.

Savings

The average monthly deposits were \$33.81 for *savers* and \$19.06 for all participants. With the matching funds, the average active participant accumulated assets in IDAs worth \$2,755.

The average savings rate was 1.6% of the monthly income.

The average net deposits for the average participant was \$528.

About 32% of participants made matched withdrawals. The money was used mainly for home purchase (28%), micro-enterprise (23%) and post-secondary education (21%). Among those planning their matched withdrawal between Jan. 1 and June 20th, 2002, their asset goals included home purchase (55%), micro-enterprise (18%) or post-secondary education (14%). The matched withdrawals in total amounted to \$662,127, or \$1,245,818 with the matching funds. The total assets withdrawn with matching funds totalled \$3,648,149.

About 64% of the participants made an unmatched withdrawal. The average amount removed without a match was \$169.

To give some perspective to these figures, the median illiquid assets — mostly homes (19% ownership) and cars (67% ownership) — were \$2950 (although skewed by one response), debt was \$2,875 and mean net worth was \$4,039.

Costs

Taken from the 2001 Evaluation Report (Schreiner et al, 2001) prepared by the Centre for Social Development (CSD), St.Louis, MO. It is based on data collected from 1997-2000. Final cost estimates will not be completed until 2003.

The average cost for running these programs (excluding the matches) was \$70.38 per participant per month, or \$2.77 per \$1 of net deposits. The costs have declined over time. The average cost was \$117.58 through mid-1999, and \$43.66 from mid-1999 to mid-2000.

These costs probably overstate the eventual cost of running these programs. They include various short-term or special costs, including those related to starting-up the programs, providing the technical assistance to other IDA programs, collecting data for the evaluation, and being involved in policy development. They also do not reflect the savings that will come as the programs grow in size and experience.

Most of the remaining and more immutable costs in these programs are associated with their service-intensive features, particularly the one-on-one involvement and counselling. These are seen to be key features in enhancing the outcomes of these programs, and reducing them might present a challenge.

b) Other Findings (Schreiner et al, 2001 and Schreiner et at 2002)

- Rate of unmatched withdrawals and exits: The size and frequency of the unmatched withdrawals, despite the loss of the matching funds, has been one of the biggest surprises in ADD. It indicates that saving is difficult for at least some participants, even in a supportive context like IDA.
- Impact of match rates: The higher match rates seem to reduce the incidence of unmatched withdrawals, but they do not seem to lead to larger deposits. There is some anecdotal evidence that suggests that higher matches also attract participants to the IDA programs.
- Impact of financial education: Financial education improves the savings but only to a point. Each hour up to 10 hours was associated with large increases in savings, but hours after that had little effect.
- Extent of asset shifts: Most IDA deposits probably came from both new savings as well assets converted from other forms, but the extent of each is unknown. There is some indication that many participants worked longer hours or tightened their household budgets to produce savings. On the other hand, because of the matches, some also may have reduced their other savings, borrowed to fund the savings or repaid debts more slowly than otherwise.
- Impact of other factors: Those with a higher education were found to save more under this program, but employment was not significantly associated with any savings outcome. Those with higher incomes saved more, but the percentage of their income saved was less than those with lower incomes.

2.2.2.Assets for Independence Demonstration

The Assets for Independence Demonstration (AFI) project was established through the Assets for Independence Act of 1998, the first federal law enacted specifically to support IDAs. The demonstration project was allocated \$125 million over 5 years to support 40-50,000 new IDAs across the country. It is operated under the auspices of the Office of Community Services of the US Department of Health and Human Services.

Under this program, grants are made on a competitive basis to non-profit community-based organizations, local governments and tribal bodies that are currently operating, or planning to implement, an IDA initiative. Federal dollars must be matched at least by an equivalent amount of local dollars.

The funds must be used within these limits:

- a maximum of 85% for savings matches;
- a minimum of 2% for data collection;
- a maximum of 5.5% for financial education; and
- a maximum of 7.5% for project administration.

The funds can be used to match savings when applied to these purposes:

- post-secondary education;
- first-home purchase; and
- business capitalization.

To be eligible, the household income must not exceed one of the three measures of poverty that are used by local organizations. Those limits are defined in the following ways:

- 200% of federal poverty level.
- The maximum for receiving assistance under the Temporary Assistance for Needy Families (TANF) program, which is part of the federal government's welfare support system.
- The maximum for receiving assistance under the Earned Income Tax Credit (EITC), which is one of the federal government's main anti-poverty programs.
- The households must also meet asset limits, and must have a "Savings Plan Agreement"
- The minimum permitted match rate relative to the savings is 1:1, and the maximum is 1:8. The maximum federal match available for each household is \$2000.
- The housing programs can be integrated with other local first-time homebuyer, leasepurchase, and construction or rehabilitation programs.
- The program was appropriated \$10 million in both of the 1999 and 2000 fiscal years. \$9.4 million was awarded to 40 organizations in 1999, and \$4.5 million to 25 organizations in 2000. \$20 million was appropriated in 2001, \$13 million for new competitive grants and \$7 million for supplemental grants to past grantees.

2.2.3. *learn*\$ave

learn\$ave is an on-going Canadian demonstration project examining the use of IDAs for adult education. SEDI conceived, designed and organized the project, and now oversees its operation. It is being funded by a \$35 million commitment from the federal government through Human Resources Development Canada.

The project is being operated through a network of local non-profit organizations in ten communities: Vancouver BC, Calgary AB, Winnipeg MB, County of Grey Bruce ON, Toronto ON, Kitchener-Waterloo ON, Montreal QB, Fredericton NB, Halifax NS, and Digby and Annapolis County NS.

The local partners are responsible for recruiting and screening the participants, providing them financial training and case management support, and monitoring the results.

The financial services are being provided by a national financial institution in nine of the communities, and by a local credit union in the other.

The results of this demonstration project will be closely monitored and evaluated by an independent organization, the Ottawa-based Social Research and Demonstration Corporation.

a) Parameters

In this project, IDAs will be made available to 3675 participants having limited incomes and assets. Specifically, their household income cannot exceed 120% of the LICOs, and their financial assets 10% of their income or a maximum of \$3000. Up to 25% of the accounts in certain communities, and up to save 75 in the others, will be available to people receiving provincial social assistance.

The participants will be able to save up to \$1500 (or \$40 per month on average) over a three-year period. The deposits will be matched generally at a rate of 3:1. This will allow for a maximum matched contribution of \$4500, and a total maximum accumulated deposit of \$6000. Financial management training, case management and peer support also will given be to most of the participants. Some the conditions, such as match rates, savings periods, and financial training, have been varied in different places in order to test their effect on savings behaviour.

The matched funds can be used only for post-secondary education, skills development, microenterprise capitalization, associated supports to learning (like computer purchase and tools of trade), and disability supports.

The project will be conducted over an eight-year period. New accounts are being opened over a two-year period starting in June 2001. The participants will be given 3 years to save. After this five-year operational period, the project will be evaluated for a further two years and completed in 2009.

The participants will be required to adhere to certain protocols, including minimum and maximum savings periods, minimum and maximum monthly savings, and restrictions on withdrawals for non-authorized uses and missed deposits.

b) Results

Data from December 2002 provides the following results:

- 1976 participants have enrolled.
- Approximately 1530 of those participants have opened an account.
- 62% of the participants are saving for education, 12% for skills training and 26% for micro-enterprise development.
- The participants have saved a total of \$555,115.82 available for matched credits.
- The participants on average have had their accounts for just over 4 months.
- The average monthly savings available for match is nearly \$55.94

Several hundred participants have taken part in financial management training courses offered by community partners. Anecdotal reports suggest that participants find the training useful, due both to the information conveyed and the peer interaction.

2.2. Other US IDA Initiatives

2.2.1. Savings for Working Families Act

This bi-partisan bill was introduced in the US Congress in early 2000. It has been passed at various times by various committees, mostly recently in mid-2002 in a modified version by the Senate Finance Committee.

As originally proposed, the act would support up to 900,000 IDA accounts, using \$1.2 billion or more in federal tax credits over nine years, and going mainly to the participating financial as a way of encouraging their funding the operations and the matches.

The financial institutions would be eligible for two tax credits:

- a 90% federal tax credit (limited to \$90 million/year for each institution) for all matching funds provided; and
- a 50% federal tax credit to help cover financial education, monitoring and administrative costs (limited to \$1.5 million/year for each institution).

Also, in separate provisions meant to encourage other community participation, a 50% tax credit would be available to any corporate or individual taxpayer investing in qualified non-profits, credit unions and community development financial institutions administering IDA programs. The funds received could be used for matches, financial education, monitoring and program administration, but at least 70% must be used for matches. The credit is capped at \$5 million/year for each taxpayer.

The bill proposes to provide a 1:1 match, up to \$500/year for each person for savings into an IDA account from earned income. Allowable uses would be first-home purchase, post-secondary education and small business start-up. Households at or below 80% of the area median income would be eligible.

The Senate Finance Committee approved a modified version of this bill as recently as mid-2002. It would provide \$450 million over nine years for the creation of up to 300,000 new IDA accounts. It also included measures to ensure that these accounts are distributed across the country, and the accounts are properly managed.

2.2.2. State-Sponsored Programs

A number of IDA programs have been created in response to state-level supports. These supports and a sample of the local programs were examined in a recent study [Abt, 2000].

a) State Supports

Many states have started to support IDAs for low-income households, generally as part of their welfare reforms. The support was initially introduced in the early 1990s through waivers given by the federal government to individual states that enabled them to experiment with welfare reforms. Later, cross-the-board flexibility was introduced through the comprehensive welfare reform legislation passed in 1996.

The 1996 legislation helped in two other important ways. It allowed states to use federal funding given to local welfare systems such as their Temporary Assistance to Needy Families (TANF) block grants, to fund IDAs. It also allowed states to remove restrictions preventing welfare recipients in accumulating assets.

By the end of 1999, 32 states had supported IDA initiatives through funding and/or enabling legislation. Out of these, however, only 7 states had established fully operational programs. These were Illinois, Indiana, Iowa, North Carolina, Ohio, Pennsylvania and Vermont.

Two states had established programs provided funding in excess of \$1 million annually. Pennsylvania provided \$1.25 million annually in 1997 and 1998, while Indiana provided \$720,000 and \$1.4 mil in 1998 and 1999 respectively.

The state funding varied widely. Most provided matching funds but some did not. Some had established permanent funding, but others only for a limited time. Some also supported administration, and others did not.

This activity as the state level did not reflect the impact of *Assets for Independence Act* of 1998, which had triggered a considerable amount of pending legislation in many states.

b) Local Programs

As part of this study, a survey was made of 16 active local IDA programs in the supportive states.

The programs, which had been operational for 8 months up to 4 years, had differing origins and characteristics. For example, they had considerably different requirements regarding the minimum size and frequency of deposits, minimum participation periods before qualifying for matching funds, and extent and nature of education and counselling.

Organizations

The local programs were typically run by local non-profit organizations that folded the IDAs into their existing operations and used them to assist the clientele that they already targeted.

Although they all generally worked within state-established guidelines as a condition of receiving IDA-targeted state funds, there were indications that they sometimes bent the rules to accommodate suitable applicants that might otherwise be ineligible.

Participants

The participants were overwhelmingly female, and a significant portion was African-American. The majority of female participants were heading households with children. The vast majority was working at least part-time.

The income restrictions placed on about half of these programs imply that almost all of their participants would be eligible for some public assistance.

Many of the states specified that the IDA deposits must be from earned income, implying that the participants must be employed. The overwhelming majority of the deposits were from earned income, both in programs that explicitly required this and those that did not.

Income Eligibility

The income eligibility requirements were based on one of these thresholds:

- a maximum of 150 or 200% of the federal poverty level; or
- 80% of the local area median income.

Eligibility for TANF also made the recipients eligible for the IDA programs in many states.

A few respondents claimed that 150 or 200% of poverty level was unrealistically low and should be raised to make IDAs more accessible to those with enough money to consider saving.

Authorized uses

Nearly all of the programs included home purchase, business start-up, and education as authorized account uses. Some also included retirement, credit repair or emergency withdrawals, home improvements, child care during education, and car purchase or repair.

Most of the local programs had a particular specialization. Those focusing on home purchase were these:

- Ft Wayne Neighborhood Housing Partnership in Ft Wayne IN;
- Muncie Homeownership and Development Center in Muncie IN;
- Affordable Housing Coalition (Asheville and Buncombe Counties) in Asheville NC;
- Passage Home (Wake County) in Raleigh NC; and
- Forsyth County Housing Program Experiment in Self Reliance in Winston-Salem NC.

Of these only the Raleigh program was exclusively for home purchase.

More than half of participants in all 16 programs were saving for homeownership.

Some of the survey respondents indicated that having a range of allowable account uses could be important because it gave participants alternatives when their original purpose — particularly, home purchase — proved difficult to achieve.

Matching Provisions

The state-funded match rates ranged from 15-25% of the clients' deposits up to 300%. Because of the additional funding raised by many of the local organizations, their match rates overall ranged from 30-50% to 400%.

In some cases, the local organizations had used the additional funds also to provide matches to populations ineligible for state money.

Some match rates varied by income level, with lower rates going to higher income levels.

2.3. Other Related US Proposals

Various other IDA proposals have been made in the US. The most notable are those made by President Bush. During the 2000 presidential campaign Bush pledged to create 1.3 million IDAs using \$1 billion in tax credits to financial institutions that operate IDAs and provide matching funds. He also cited IDAs in his 2000 State of the Union address.

In related provisions, the president more recently has proposed a program supporting "retirement savings accounts". Funded by a proposed \$54 billion over 10 years, it would provide matching contributions for savings made for retirement. The match rates would start at 2:1 for families earning less than \$25,000, and be phased out at \$80,000. Although focused on retirement, after five years of savings the provisions would allow pre-retirement withdrawals for first homes and college or medical expenses. The program would provide no formal role for non-profits, financial education, or the private sector. It would be financed through tax credits to the financial institutions and employers.

2.4. Canadian Programs

There are two recent IDA programs in Canada, one in Calgary and the other in Winnipeg, that promote savings for home purchase.

2.4.1. Calgary's Fair Gains & Owen Hart Homeowners Program

These two IDA programs have been developed and are operated by MCC Employment Development in Calgary. MCC is non-profit agency of the Mennonite Central Committee that was established in 1991 to address the needs of low-income people in Calgary. As part of that mandate, it also operates a security deposit program and is a Community Partner administering learn\$ave.

Fair Gains, started in 1999, combines developing money management skills with saving for asset development. The savings can be used for training or education toward a career, education for a child, starting and expanding a business, and purchase of a home.

The Owen Hart Homeowners Program, started in 2000, is devoted to preparing the participants for and saving for homeownership. In order to qualify for this program, the participants must pass through Fair Gains.

United Way is a partner with MCC in both programs.

a) Fair Gains

The participants are required over one year to deposit a minimum of \$15, and allowed to deposit a maximum of \$45 per month. After one year, their savings are matched at a ratio of 3:1 toward asset development.

Their minimum savings in the year would amount to \$200, which would be matched by another \$600. Their maximum savings would be \$600, matched by another \$1800.

Over that year, in order to develop their money management skills through, the participants are expected to attend regular workshops, peer group meetings and one-on-one consultative sessions. This process involves each in about 30 hours of educations, and costs the organization about \$2000 per participant.

Participation is limited to those with an income below LICO, and with minimal savings and investments.

For the first three years, the program took in 20 participants per year. This year that number has been increased to 40. Roughly, about 3/4 of the participants enter the program with the intention of buying a home, but only about 1/4 of the total enter the Hart program. About 1/4 drop out of Fair Gains, and about 1/2 use their savings and matches for another purpose. Many of the latter find home purchase is premature or inappropriate for them personally, mainly because of insufficient income.

b) Owen Hart Homeowners Program

Under this program, the participants over one year are required to save a minimum of \$45, and allowed a maximum of \$90. These savings at the end of year are matched at a 4:1 ratio.

They are also expected attend about 10 hours of education.

Combining the savings and matches from the two programs, the participants at the end of two years can save a minimum of \$740 (then matched by \$2,760, for a total of \$3,500) and a maximum of \$1,680 which is matched by \$6,120, for a total of \$7,800.

So far under this program, all five participants from the first year have purchased homes. Five more are in the program and another five will be starting shortly. Of the five purchasers, 3 qualified for a mortgage (one through a co-signer) and 2 assumed existing mortgages.

The participants in the program so far have saved at or near the maximum. This can be ascribed to various reasons. The selection to the second stage is limited to those most able to take advantage of it; and by that time, the participants have "learned the tricks" to saving.

In addition, although the incomes of the initial participants are at or under LICO, virtually all of those that continue into this program have had an increase in income. Those at LICO probably would not be able to qualify for a mortgage.

Observations

The high match rates are seen to be important for attracting people to the program and reducing attrition, but the relation developed between providers and participants over time is also very important their completing home purchase because of the time and obstacles involved in the process.

By having the year-long education program prior to deciding on homeownership, the providers are able to select with some certainty the candidates that will succeed. The merit of not having the savings in the initial program tied exclusively to homeownership means that those participants that find home purchase inappropriate are still encouraged to continue saving because they have alternative ways to use their accumulated assets.

2.4.2. Winnipeg's Housing IDA Program

Winnipeg's IDA program was operationally started in October 2000 by the Alternative Financial Services Committee, a partnership of five organizations: the Assiniboine Credit Union, Mennonite Central Committee, North End Community Ministry, SEED Winnipeg, and United Church Conference of Manitoba and NW Ontario. The North End Community Ministry is responsible for operating this program as well as participating as a Community Partner in the *learn\$ave* project.

The committee was established in 1996 in response to the recognized need for accessible and affordable financial services as well as money management training for low-income families. For example, the committee now offers free tax preparation services to low-income residents, and has been investigating other alternative financial services like pay-day loans, cheque cashing and consumer lending circles.

Funding, amounting \$250-300,000 for a three-year contract, was provided by the province, United Way and local foundations and anonymous donors.

a) Project Conditions

This IDA program is based on the following parameters:

• A match rate of 3:1

- Minimum savings of \$15 per month and a maximum of \$250. (The original maximum was \$45, but this was raised to be consistent with *learn\$ave*.)
- Maximum saving period of 3 years.

It is open to all low-income residents of Winnipeg, but recruiting focuses on those in the Inner City and North End, the two most distressed areas of the city.

The maximum eligible household income is 120% of the LICO. This maximum is considerable reasonable for qualifying for a mortgage loan to buy a house in Winnipeg.

Money saved by the participants is be matched by contributions to a parallel account on a 3:1 ratio. Participants are not able to withdraw money from their matched account without prior approval.

The matched funds can be used for the purchase or renovation of a home. When initially conceived, the program also was to include investments in small businesses as well as education and training. This was changed with the advent of the *learn*\$ave program.

All participants are required to attend a comprehensive money management program, which is provided through seven workshops over 2 months. They also are provided with one-on-one counselling.

b) Participation

The one-time intake (December 2000) for this program was 40 participants. To date three participants have purchased homes³ and a number of participants have cashed out partially or fully for home renovations. The average monthly savings has been \$75.

These participants have saved over 1½ years on average about \$500, or \$25-30 per month. At this rate, over the full three years, they each will save \$1,000, which will be matched by another \$3,000 from program. \$4,000 is considered to be an effective downpayment for a house in Winnipeg.

2.5. Local US Programs

This section summarizes the findings of a series of interviews conducted with administrators of local IDA programs in the US that target exclusively or mainly homeownership. The interviews were conducted by telephone in September-October 2002.

2.5.1. Methodology

The programs were identified through two sources:

³ According to Andrew Douglas, IDA Coordinator, Northend Stella Community Ministry/Seed Winnipeg, the homes purchased ranged from \$45,000 to 79,000 and the participant's incomes ranged from just under \$20,000 to \$25,000.

- a previous survey of a sample of local IDA programs [Abt, 2000] that identified a small number of homeownership IDA programs (see section 2.5); and
- the *IDA network, a* website that identifies about 400 IDA programs across 44 states, but does not distinguish those targeting on homeownership. To find relevant organizations, a search was made for programs that had 'housing' or 'homeownership' in their name in the largest states, and those generally most active in supporting affordable housing.

A total of 23 programs in 14 states were identified. Interviews were conducted with operators for 13 of these programs. Limited information was obtained on another program, which was temporarily suspended while being re-organized. Two other programs were found to be defunct.

The interviews were held with the administrator or manager of the IDA program, or failing that, head of the agency or the manager of the education and counselling. In many cases, more than one person was interviewed. When available, information was taken from their websites and/or faxed brochures.

The following provides the jurisdictions and names of these organizations:

- Columbus OH: Columbus Housing Partnership
- Montgomery MD: Montgomery Housing Partnership
- Westmount County NJ: Housing and Economic Opportunity Inc
- West Sacramento CA: Mercy Housing CA
- Bunscombe County and City of Asheville NC: Affordable Housing Corporation
- Medina OH: Medina Metropolitan Housing Authority
- Forsyth County NC: Experiment is Self-Reliance
- Ft Wayne IN: Ft Wayne Neighborhood Housing Partnership
- DuPage IL: DuPage Homeownership Center
- Cape Cod MA: Housing Assistance Corporation of Cape Cod
- Bremerton WA: Bremerton Housing Authority
- Denver CO: Hope Communities Inc, El Norte Neighborhood Development Corporation, Rocky Mountain Mutual Housing Association, and Mile High United Way

2.5.2. Organizations

All of the organizations are non-profit, none are government agencies but two are independent housing authorities established and funded by government.

a) Background in Affordable Housing

All but one of the organizations were originally involved in some way in providing, managing or supporting affordable housing.

The single exception is an organization that is involved primarily in asset-building. It is part of the nationwide network of Community Action Agencies, and operates a broad range of programs directed at helping lower-income people to become socially and economically self-sufficient.

All of the others have their roots initially in one of the following fundamental activities:

- building or rehabilitating affordable housing, mostly for rent but sometime for homeownership, and mainly in run-down urban areas;
- operators of homeownership centers, which promote homeownership for lower-income households through advice, referrals, education and support services, but do not develop housing;
- providers of housing assistance, such as downpayment assistance, "soft" second mortgages, low-interest financing and rental housing subsidies; and
- managers of non-profit rental housing.

Having started in one activity, many of these organizations over time have extended the range of their activities into other fields, including homebuyer education and counselling.

Three of the IDA programs were started by organizations managing non-profit rental properties. The smallest of these properties has 50 units, and largest has 800 units. (In one case, the organization is one of three local non-profit housing owners that jointly run the program.) The programs have been established solely for the benefit of their tenants. They are seen as a way of assisting their tenants build equity, and perhaps buy a house. They potentially have the second benefit of freeing up rental units.

b) Consortia

Some of the organizations run the IDA programs through a collective arrangement and basically have formed a consortia for operational purposes..

In one program, the organization leading the program operation is part of a long-standing partnership of 13 non-profit agencies that offers a housing range of housing-related and community-based services. In this case, the education and counselling, and also the business development and adult learning components, are handled by other organizations within the partnership.

In three other programs, a consortium of existing agencies was established to operate the IDA program collectively. In the cases, the agencies typically had different roles — intake, education and counselling, accounting, fund-raising and so on.

The latter approach has been taken by the largest organization interviewed, where it was considered a successful working model. It was also used by one of the organizations winding

down, and where it was not considered successful. The difference in the experience of the two largely lies in how well they kept the expertise and roles from overlapping.

In two of these programs, this approach seemed to be used so that the direct contact with participants could be handled by local agencies in the separate and disparate communities, but the overall administration and accounting as well as the education and counselling could be handled centrally.

c) Experience with Education and Counselling

Most of these organizations had previously provided education and counselling — in one notable case for nearly 20 years, and others for 10 or more years, before they adopted IDAs.

These organizations started using IDAs as an extension of the services they provided to their existing clientele. For example, organizations providing affordable housing or housing assistance for lower-income homebuyers started education and counselling after too many of these buyers defaulted on their mortgages.

One group of organizations had no previous experience with education and counselling. These are the managers of rental housing. Two established their education and counselling services as part of their IDA programs, while one relies upon another existing education and counselling program provided by another agency. One other small organization also relies upon another existing program.

In most cases, these courses are open to the general public. They are not exclusively for the IDA participants, nor are they even exclusively for lower-income households. In some case, priority is given to potential IDA participants where there is a waiting list.

Most of the programs break down their educational courses into distinct components. The participants typically start with financial literacy training. Following that, the participants are streamed into separate courses devoted either to first-time home buying, or higher education or business development if appropriate. Sometimes, these different courses are given by separate organizations that specialize in those subjects.

d) Permanence

Only roughly half of these IDA programs can be presently considered permanent and on-going programs. The others were variously described as trial or temporary programs or programs with one-time funding. Two of these were presently winding down. One had been temporarily extended, and three others might be temporarily extended. As noted, another had been suspended and its future was uncertain.

Many of the organizations had changed over their brief existence. Although not certain in all cases, the main reason seems to revolve around financing, either the initial funding ran out and/or the costs of operating the program were higher than expected.

2.5.3. Uses for Savings

Only 5 of the programs are devoted exclusively to providing for homeownership. The remainder allow for two or three alternative uses. These include higher education (7 examples), business development (5), car purchase (2) and home repair (1).

In the one example, home repairs are limited to certain prescribed essential repairs, and not "cosmetics". The use of the savings is subject to obtaining three quotes for the work, and getting prior approval of the IDA administrator.

In some programs, the other uses are included mainly to give the participants an alternative if they were unable to buy a home. In other programs, they are the result of requirements imposed by a major funding source.

Even in those programs with alternative uses, most or nearly all of the participants are saving for homeownership.

All of the programs allow the matched savings to be used for downpayments toward home purchase and also for the associated closing costs. Indeed, because other downpayment assistance and favourable mortgage terms are often available, most or all of the savings in many programs are used for the closing costs only. (Mortgages with 3% downpayments appear to be available in all of these jurisdictions, and mortgages with 0% downpayments for some homebuyers.)

Furthermore, in one notable case, because of the other downpayment and closing cost assistance available to all participants, the savings in one program have been used often to reduce the mortgage principal.

Some of the program have either expanded or contracted the options. Two programs that initially allowed for education and micro-business as well as homeownership are now focusing only on homeownership, one because its latest funding source required this, and the other because it considered homeownership to be the greatest need. Conversely, one program initially focusing solely on homeownership has expanded its options due to its latest funding source, and another is considering the same in order to provide more flexibility for its participants.

There was very mixed response when the operators of homeownership-only programs were asked about what would happen to those participants that could not buy a home. At least two operators felt that this would be unfortunate, but also felt the participants would still benefit considerably from the education and counselling aspects of the program. At least one, as indicated above, is looking into adding an alternative use for these participants. Two others felt that solution lies in proper screening so that only those households likely to succeed were given entry into the program. (As noted later, other operators were concerned about attempting to screen in this way.)

2.5.4. Details of Programs

The largest of these programs has had 420 accounts, including both active and cashed-out accounts. For the other organization, there are three with roughly 150, 100 and 75 accounts respectively; four with 25-50; and five with 5-25.

Some 165 participants had purchased homes in the largest of these programs. All of the remainder together had supported so far roughly about 75 home purchases.

Six of the programs set their maximum matchable savings at \$1000. The other maximums are \$600, \$800, \$900, \$1200, \$1320, \$1650, and \$2000.

With the maximum matches available, the total potential matches are \$2400 (2 examples), \$2700 (1), \$3000 (3), \$5000 (5) and \$6000 (1).

The maximum savings periods range from 2 to 3 years, with the majority at 2 years. The main exception is one program that allowed the savings to be made over a 10-month period.

a) Savings Rates

Most of the programs allowed for maximum saving rates of \$40-55 per month on average. The two highest rates were \$100 and \$70.

At least two programs place a limit on one-time deposits as a way of encouraging consistent deposits over a long term, and discouraging participants in taking advantage of tax rebates. These limits are set at \$250 and \$150.

In virtually all of these programs, strong emphasis is placed on making consistent savings over a long period of time. This consistency was seen as evidence of that the participants were taking control of their household budgets, and would become good credit-worthy customers for mortgage loans. As noted later, the amount of the monthly savings would have agreed by the participants at the outset of the program.

One small incipient program stringently requires the participants to make all monthly payments. Any failure is considered grounds for removing the participant from the program.

Virtually all of the other programs are more tolerant. They generally will allow the participants to miss one or two consecutive months, but they will expect them to provide a reasonable explanation and make up the missing amounts. Missing a payment in some programs will trigger a counselling session.

b) Match Rates

The basic match rates for these programs range from 1:2 up to 1:4.

One notable anomaly is the 1/3:1 match rate used by one of the tenant programs. (In other words, for a maximum saving of \$1000, it provides a match of \$333.) The program expects to increase this to 1:1 in the near future.

At least two of these programs provided different match rates for residents of different areas or income levels. These differences are the result of the criteria imposed by different funding sources.

One of the effects of the different criteria and match rates is accounting complexity. One administrator noted that this added a considerable administrative burden.

Two of the programs have been given funding which they can at their discretion to increase the match rate from 1:2 up to 1:8 and 1:4 respectively for participants with special needs.

Four of the programs have just or are expecting to increase their match rates. In all cases, the reason given was the need to provide more assistance in the face of high and/or rapidly rising house prices. (In at least two of these jurisdictions, the house prices had increased over 50% within the short period of the program.) One will be increasing the match from 1:3 to 1:6.

c) Eligibility Requirements

Five of the programs use 80% of local area median income (AMI) as the upper income threshold. Four use 200% of the poverty level, which seems to translate to 50%-70% of AMI in the different jurisdictions. Three respectively use 65%, 60% and 50% of AMI. Although not certain, the income thresholds appear to have been imposed by the funding sources.

There is one notable exception. One program has no income threshold because it operates in a revitalization area, where homeowners with a mix of incomes are being sought.

One program lowered its threshold from 80% of AMI to 200% of the poverty level due to changes in its principal funding source. As a consequence, it noted a significant increase in the participants needing more support and assistance to achieve homeownership.

Some of the programs also use various other requirements. Many require the participants to reside within the local jurisdiction. Three, as indicated earlier, require them to be tenants in their housing developments. Some require them to be employed. One program requires them to be on welfare.

d) Admission Process

The IDA participants must complete a prescribed educational and counselling program before being accepted in the IDA program.

Before starting the IDA program, the participants generally are also interviewed. At this time, their income and assets are checked, and their credit problems, work history, and household budgets typically are also reviewed. In most programs, the participants then are also expected to set monthly savings targets for themselves, and to sign a contract committing to those targets.

Two of the IDA programs rigorously screen the potential participants in order to accept only those likely to be able to buy a home.

Most of the programs do not pre-screen their participants in this way. Indeed, two administrators felt that it would very difficult to identify those participants likely to fail or succeed. One was also concerned about the potential for prejudice possibly affecting the decision.

2.5.5. Sources of Funding

Funding for these programs came from wide variety of sources. Four have one funding sources; in three of those cases, it is from the state or county. Most rely on multiple sources. Many have changed their funding sources over time.

Four programs receive funding from the *Assets for Independence* Demonstration (see section 2.2). One of the conditions of receiving this funding is that it must be matched at least by equivalent local funding. There were numerous complaints about this funding source; it was considered to inadequately support operational needs, and to impose restrictive eligibility requirements.

Three programs use funding from the Federal Home Loan Banking system. This was considered to be a relatively generous and flexible source of funding. (The background to this funding source was not pursued, but it apparently comes out of the S&L clean-up. As a result, banks in FHLB system are now required to put aside 2% of their profits into an affordable housing funding pool, which support IDA programs and other special projects.)

The funding has also come from all levels of local government, including states (6 examples), counties (5) and cities (3). Although not known in all cases, all or most of this funding was likely to have originated as federal block grants either for housing (through the HOME or CDBG programs) or welfare (under the TANF program).

Local community or family foundations provided funding in four of the programs. The amounts were relatively small and one-time for "capacity-building". (This is consistent with the practice of most US foundations, which generally do not engage in on-going assistance.)

Other singular sources of funding include these:

- Neighborhood Reinvestment Corporation and Enterprise Foundation, which are two
 national intermediaries that typically provide "capacity-building" support to help in
 starting community-based organizations or initiatives, but not on-going support;
- A state non-profit capital corporation, which uses tax credits to finance the affordable rental housing and has established a funding pool for special projects for the tenants of these projects;
- A special state fund using one-time surplus welfare monies resulting from recent welfare reforms:
- The local board of realtors; and
- The local mortgage bankers association.

The banks are not significant Funders. They provided money in only two cases and in one case, it was described as a "stipend".

The United Way is a fund-raising partner in two of these programs.

There were various comments about funding for matches being easier to secure than funding for operations. There were also other indications that the operational aspects were under funded.

2.5.6. Use of Housing Assistance

Nearly all of the administrators expected that most of the participants would utilize some form of other government assistance in purchasing a home. While this was not always necessary for households earning 80% of the area median income, the need for additional assistance increased for households with more limited incomes.

Two of the administrators felt that they had an obligation to ensure that participants received housing assistance if needed to achieve homeownership. Most of the others only expected to make participants aware of options and make referrals for them, but not to be involved in securing that assistance.

About half of the organizations are either directly involved in providing some form of housing assistance, or in a consortium with other partners that do.

The most common forms of housing assistance appear to be interest-free deferred second mortgages and downpayment assistance in the form of forgivable loans. One or both appeared to be available in a majority of the jurisdictions where the IDA programs were operating. New or rehabilitated ownership, provided at a reduced price through government assistance, also appears to be available in some areas, particularly as part of revitalization programs.

In at least three jurisdictions, low-interest financing is available for mortgages from a funding pool supported by a number of local banks.

In another recently approved and potentially important source of assistance, tenants receiving rental assistance from the federal government through Section 8 vouchers are now permitted to use that assistance toward purchasing a home.

2.5.7. Involvement of Banks

All of the organizations have established an effective relation with one, and sometimes two or more banks, that hold the accounts. One administrator, however, did note having considerable difficulty in finding an interested bank.

In general, the banks waive service charges and minimum deposit requirements. A few have trained their tellers and staff to deal with the IDA participants. At least one had nominated and trained one staff member to handle any special problems.

In many of the programs, the lenders provide mortgage pre-qualification for the IDA participants as part of one of educational or counselling sessions. This enables the participants to determine if they are likely to get a mortgage before they start the savings program, and if necessary, to address any credit or other problems.

Outside of these activities, the involvement of the banks is relatively limited. In some cases, bank representatives participate in the homebuyer or literacy courses, particularly to explain the lending criteria and various mortgage products. In two cases, as noted, the banks have provided some funding support. In two cases, the bank representatives sit on the advisory board for the program or one of the organizations.

In all of these programs, the banks do the underwriting for the mortgage loans. One of the IDA organizations was asked to undertake this service by a local bank, but refused as it did not want to associate the program with one potential lender. (Another one of these organizations does the underwriting for a separate low-interest financing program funded by a consortium of local banks.)

Some of the programs have intentionally limited the participation of the banks. The administrators are concerned about associating the program with any single institution. They see promoting any single bank as contrary to training their participants to become knowledgeable consumers and to seek the best deal for themselves.

The reason for the bank involvement, as described by many of the operators, is that they want the additional business, not only for the mortgage loans but also car loans and other needs. Some of the operators felt that the Community Re-investment Act also was likely to be an important factor. (The bank's assistance would be a positive factor whenever their bank charter was subject to review.)

Many of operators, nevertheless, said that the bank also had to have some understanding and sympathy for purpose of the program.

Many of the programs use so-called "custodial accounts", which gave them control over the deposits. As a consequence, they also receive the monthly statements that can be used to monitor the savings. Two administrators, however, questioned the appropriateness of using these type of accounts in a program that is directed at developing self-sufficiency.

2.5.8. Monitoring

Most of the programs do not monitor the participants after they complete their education and counselling programs, nor their IDA programs. Only two of the programs undertake post-purchase monitoring and pro-actively intervene with counselling when there is a risk of mortgage default.

Many of the organizations have tried to monitor but stated that they do not have adequate operational funding. Post-purchase monitoring also was considered to be particularly difficult and time-consuming, especially as the participants do not readily respond to surveys or enquiries.

Many of the organization provide default prevention counselling, but only when the participants volunteer to come forward. In general, this was not considered adequate because the participants often waited too long before seeking help.

At least two organizations offer post-purchase education courses. These deal with such aspects as maintenance, budgeting and predatory lending practices.

2.5.9. Impact on Mortgage Defaults

Many of these organizations, even though they do not monitor the participants, have hard evidence about the impact of homebuyer education and counselling on mortgage defaults. As noted earlier, the organizations started education and counselling services after having operated various other programs assisting homeownership, such as selling units and providing loans. The impact of these services was to reduce the number of mortgage defaults in their programs from an excess to none/nearly none.

All of the other administrators associated with long-standing education and counselling programs, but which did not have corresponding direct evidence, also strongly felt that these programs significantly reduced default rates. Most also said that the participating banks also agreed with that view.

Because of the more limited experience with IDA programs, less information is available on their impact on reducing defaults specifically, or changing behaviour generally.

While this question was not specifically asked, two of the administrators indicated that they had more confidence in the effectiveness of education and counselling than in IDAs as a way of supporting homeownership by lower-income households. This could be taken to reflect their longer involvement with the education and counselling than IDAs.

Two administrators also emphasized that the peer involvement and support are also particularly important for the outcome of these programs.

2.5.10. Special Programs

Two special IDA-type programs were identified in the course of the interviews.

One is an employee-assisted IDA program, which now involves a hospital and is expected to include a school board, a major tourist-based industry and also the municipality. The hospital has agreed to provide a 2:1 match for savings up to \$1250 made by employees toward downpayment. It is seen as an employee retention program in a high-priced area where employees have difficulty buying homes.

The other is part a downpayment program being started by a local bank. It will provide a 2:1 match for up to \$960 in savings as insurance against mortgage default. The savings will be disbursed when there is a major crisis that interrupts mortgage payments.

3. CANADIAN GOVERNMENT HOMEOWNERSHIP PROGRAMS

This section describes the current federal, provincial and municipal government programs that support home purchase in Canada.

3.1. Federal Programs

3.1.1. RRSP Home Buyers' Plan

This program is a provision of the federal Registered Retirement Savings Plan. It allows those with registered retirement plans to withdraw money from them in order to assist in buying or building a home. The withdrawn money is not taxed, provided it is repaid in equivalent installments over a maximum of 15 years starting in the second year after the withdrawal.

The plan is available to individuals or couples who have not owned a home that has been occupied as a principal place of residence in any of the past five calendar years. Each spouse having a registered plan is able to withdraw up to \$20,000, for a maximum total of \$40,000 per couple.

Both existing and newly built homes are eligible, provided they are located in Canada, and occupied as a principal place of residence within one year after buying or building it. Shares in co-operative housing corporations also qualify.

The plan was introduced by the federal government in 1992. The first-time buyer restriction was added in 1994. Since 1999, the plan has been opened to certain existing homeowners namely, previous and fully paid-up users of the plan purchasing a home designed for disabilities for themselves or a dependent relative.

The plan has proven to be a viable method of helping first-time home buyers to generate enough savings to afford to purchase a home. Since its introduction, according to data reported by CMHC, over one million Canadians, amounting to one third of first-time buyers have used the program.¹ According to CCRA figures reported by CMHC, the withdrawals have averaged nearly \$10,000.²

²Ibid

¹ "Home Buyers' Plan Helps Turn Homeownership Dream into Reality"; CMHC Housing Facts; Vol. 5, No 4; 10 April 2000 and "Home Buyers' Plan Turns Home Ownership Dream Into Reality"; CMHC New Release, 29 March 2001.

Can be found at: http://www.cmhc-schl.gc.ca/en/News/nere/2001/2001-03-29-1400.cfm

The RRSPs, like the other tax deferral savings programs, are not targeted to the poor. Because they promote individual savings through deductions to income tax, these vehicles are designed for and primarily used by middle- and upper-income households in the higher marginal tax brackets.

3.1.2. CMHC's 5% Down Program

This program (formerly called *First Home Loan Insurance Program*) was opened in 1992 to assist first-time homebuyers in obtain financing for purchasing a home when they have sufficient income to carry the mortgage debt, but insufficient savings or other resources for making the conventional downpayment of 25% of the sales price. The program limits the risk to the mortgage lenders by providing default insurance for mortgages secured with as little as a 5% downpayment.³

The insurance premium paid by the homebuyer depends on the amount of the loan as a percentage of the lending value (known as the loan to value ratio). For example, for a \$200,000 mortgage loan secured with a 5% downpayment, the cost of the premium is \$7,500. The premium may be added to the mortgage (see table on next page).

This program was created as a temporary measure, but was extended twice and then made permanent in mid-1998.

Other changes were introduced in mid-1998. The program is now no longer limited to first-time buyers; repeat home purchasers are now eligible under certain conditions. The premium for a insurance covering mortgages over 90% was also raised to 3.75% from 2.5%. The other rates stay the same.

There are restrictions on the house prices and maximum gross debt service ratios as a way of ensuring the homebuyers do not over extend themselves. The price ceiling is currently \$300,000 for homes located in the City of Toronto and Vancouver areas, \$250,000 in various other high-priced centres as well as northern areas and \$125,000 to \$175,000 elsewhere. The ceilings are subject to review and change.

Under the current program maximums, buyers may use up to 32% of their gross family income for payments of the principal, interest, property taxes and heating, and the total debt loan cannot exceed 40% of the family income. Their downpayment must come from their own resources, such as savings, sale of investment and family gifts that need not be repaid.

³Similar default insurance on low-downpayment mortgages, or high ratio mortgages, is now also offered by GE Capital Mortgage Insurance Canada.

CMHC Premium Schedule

Loan-to-Value Ratio	Premium (% of loan)
Up to 65%	0.50%
75%	0.75%
80%	1.25%
85%	2.00%
90%	2.50%
95%	3.75%

[Note: These rates are for a single advance. Progress advances are subject to a premium surcharge of 0.5%.]

According to CMHC, in the first year more than 64,000 households used the program to purchase a home. This represented approximately 1/3 of CMHC's total homeowner insured loans in this period. As of mid-1998, 610,000 Canadians have purchased their first home using a 5% downpayment. About 70% of those buyers could not have purchased their homes without this program.⁴

3.2. Provincial Programs

3.2.1. Ontario Home Ownership Savings Plan

The Ontario Home Ownership Savings Plan (OHOSP) was established to help lower-income first-time homebuyers in purchasing a home. (Two similar programs in Quebec and Nova Scotia have been terminated.)

The plan provides assistance in the form of a refundable tax credit for savings made in an account dedicated to this purpose. The tax credit is based on the amount deposited into the plan each year, and household income for that year (see table on next page). It is provided for annual contributions of up to \$2,000 per person, or \$4,000 per couple. The maximum that can be claimed annually is \$500 per person or \$1000 per couple. That amount declines for persons or couples with higher incomes, or making smaller deposits.

⁴Toronto Real Estate Board: "RRSP Home Buyers' Plan"; May 1998. For more information please see: http://www.mls.ca/boards/treb/gov_prog/rrsp.htm (this link will give you general info on how to use RRSP's toward the purchase of a home).

Tax Credit available for a \$2000 deposit under OHOSP

Net Income	Tax Credit
Individua	als
Up to \$20,000	\$ 500
\$25,000	\$ 375
\$30,000	\$ 250
\$35,000	\$ 125
\$40,000 and over	\$ 0
Couple	s
Up to \$40,000	\$1000
\$50,000	\$ 750
\$60,000	\$ 500
\$70,000	\$ 250
\$80,000 and over	\$ 0

The plan is available for individual earning less than \$40,000, or couples jointly earning less than \$80,000. The participants also must be Ontario residents and 18 years or older; have never owned an eligible home; nor used OHOSP before.

The tax credits can be claimed in each of the first five calendar years of the plan. The funds must be used, and the plan closed, within seven years.

The plans are administered by participating financial institutions across the province. The claims are made through the annual federal income tax returns.

A refund for first-time purchasers of the Land Transfer Tax, which was a component of the original OHOSP program, has not been continued.

OHOSP has been used by more than 250,000 homebuyers in Ontario.⁵

⁵ Ottawa Real Estate Board: Ontario Home Ownership Savings Plan, 2002. Article available at: http://orebweb1.oreb.ca/frm_info.html

3.2.2. New Brunswick's Home Ownership Program

This program offers financial assistance to low and modest-income families buying or building a modest first-time home.

It is available to families and individuals resident in the province, and who are first-time homebuyers or living in a substandard housing unit. Their total income must be below \$35,000, and they must be able to secure private financing and have a 5% downpayment for home purchase.

The maximum income threshold has been adjusted over time. The most recent has been determined through consultations with credit unions, and attempts to separate those households that should be able to affordable market housing without the assistance from those that cannot.

The assistance is available only for housing meeting certain criteria regarding "modest" size and amenities. No explicit limit is placed upon the price of the house.

The assistance is in the form of a repayable loan amortized over a period not exceeding 25 years, and subject to terms of one year.

When purchasing an existing unit, the loan is limited 20% of the purchase price of the unit. The amount of assistance available is based on a sliding income scale that is a percentage of the purchase price.

When building a new unit, the loan is limited to 50% of the total construction costs up to a maximum of \$40,000.

In both cases, households earning less than \$20,000 will be charged no interest on the loan. For each \$1000 of income above \$20,000, the interest rate will be increase by 0.5% until it equals the provincial borrowing rate. The household income is assessed annually, and interest rate adjusted accordingly.

3.2.3. Saskatchewan's Neighbourhood Home Ownership Plan

Saskatchewan introduced this program in 1998 to provide housing assistance for families with limited incomes when purchasing homes in designated inner-city areas within four cities.

The funding is limited to families with dependent children, and earning a maximum gross income of \$30,000 annually. They also must be capable of qualifying for a conventional mortgage for the balance of the appraised value of the house.

The assistance is provided principally through an interest-free equity loan of 30% of the appraised value of the home up to a maximum amount of \$13,500. The provincial government is responsible for up to \$11,250 and the participating municipality for up to \$2,250. A renovation grant is also available for any emergency repairs.

The loan generally must be repaid over a maximum of a ten-year period. Depending upon family income, all or a portion of the equity loan might be forgiven at the end of five years or extended at no-interest for a longer period.

The program is being operated through Quint Housing Co-op in Saskatoon, New Beginnings Co-op in Prince Albert, Hope Housing Co-op in Regina and a similar organization in Moose Jaw.

The families must initially form and work together through a housing cooperative. The housing co-operative purchases the property and holds the mortgage on an interim basis for a minimum of five years. The renovations are organized through the co-operative, using the equity loans and renovation grants on a pooled basis. During this period, each of the families leases their units from the co-operative, and is expected to help in renovating and maintaining the property. After that time, they are able to assume the mortgage and title of their home, subject to fulfilling all of the program requirements.

This program makes housing more affordable by providing no-interest short-term loans that serve to reduce the amount of money that must be borrowed through a mortgage. This, in effect, reduces the minimum income threshold for home purchase.

3.3. Municipal Programs

3.3.1. Surrey's Home Ownership Assistance Program

Surrey BC started this temporary program in mid-2002 in order to financially assist families not able to buy a home due to their limited income. The assistance is for first-time buyers living in Surrey and having a gross annual income 10% below the city's average household income. That translates to a maximum of \$49,000 in 2002.

Assistance will be provided through an interest-free loan toward the purchase of a newly constructed home located in Surrey. The amount of the loan will be determined by the lesser of \$15,000 or 15% of the market price of the home. The assistance must go toward reducing the first mortgage, and not the downpayment.

The loan will be registered on title as a 'silent' second mortgage. Up to 50% of the second mortgage will be incrementally forgiven over the first 5 years of occupancy at a rate of 10% per year, provided the owners continue to reside there. The remainder will be recovered after 20 years, or upon resale of the home or if the owners cease occupancy.

3.3.2. Winnipeg's Rent-to-Own Program

This program (details of which have been supplied verbally by Linda Ring, RRAP Co-ordinator for the City of Winnipeg), is supported by the three levels of government through grants of up to \$18,000/unit from the federal Rental RRAP Program, and \$10,000 from both the province and the city. The assistance is available only in certain inner-city revitalization areas. It is being currently provided through two community-based non-profit development organizations, which are responsible for purchasing and renovating existing houses, and for renting and then selling them to income-eligible tenants.

The non-profit organizations must retain title for five years after renovation. (This represents a reduction to the conventional 15-year RRAP requirement.) During that period, the rent paid by the tenant goes towards paying the property taxes, insurance and mortgage. The rent paid during this period goes to an equity build-up used as the downpayment for home purchase if desired. After that period, the tenant has the option to purchase the house for the initial purchase price, prior to improvements, equal to the outstanding balance of the of the first mortgage.

When the unit is sold after five years, the non-profit organization will hold, for a further ten years, a silent second mortgage, for the value of the government improvement grants. The value of these grants will be forgiven incrementally over that time. The second mortgage also will give the non-profit organizations the first right to re-purchase the unit during that time.

The tenants are limited to households earning within the RRAP guidelines. The rents are set by the Housing Income Limits determined annually by CMHC and based upon the median of rental market for the city.

3.3.3. Winnipeg's Housing Opportunity Partnership

The Housing Opportunity Partnership (HOP) was established as a privately non-profit organization in late 1997, in large part due to the efforts of the Winnipeg Real Estate Board (WREB). Its purpose is to support homeownership in the rundown inner-city neighbourhoods of Winnipeg as part of wider efforts promoting neighbourhood improvement and stability.

HOP established a revolving fund initially through contributions from two sources:

- \$600,000 over four years provided by the provincial government from the interest earned on real estate broker accounts; and
- \$500,000 over three years from the Home Equity Program established under the Winnipeg Development Agreement.

Support also has been provided from other sources. Members of WREB contributed \$25,000 in initial start-up funding to HOP, and they also provide real estate services and other assistance at no charge. The Winnipeg Housing and Homelessness Initiative provided \$200,000 in additional funding in mid-2002.

The fund is used to purchase and renovate existing houses in the inner city area. After renovation, the units are sold at a market-rate, with intent of at least recovering its costs. The resale proceeds are returned to the fund, and used again in the same way.

Contrary to its intent, HOP has been losing money upon the resale of these units. It has been recently selling the renovated homes, worth approximately \$80,000, after the comprehensive renovation (and allowing for various interim expenditures like administration, utilities and taxes), for about \$70,000. The market gap is approximately \$10,000 per unit. This market gap has been financed by contributions from the Winnipeg Housing and Homelessness Initiative [Drdla 2000].

The homebuyers must meet NHA requirements under CMHC's mortgage insurance program and have the necessary income to qualify for a 95% mortgage loan. Applicants have been encouraged to contribute towards the downpayment through sweat equity, which could count for up to half of the minimum 5% requirement. This provision of the program has been difficult to administer and seldom used. As a consequence, HOP has recently arranged to provide financial assistance out of its fund for downpayments also for up to half the minimum requirement. The assistance will be limited to households earning less than \$42,500 [Drdla 2000].

3.3.4. Montreal's Domi-cible Program

The city along with the provincial housing corporation, the Société d'habitation du Québec (SHQ), initiated this program in February 2001. Its purpose is to support the purchase and renovation of the existing plexes in targeted central neighbourhoods surrounding the downtown core [Wexler, 2002].

This program, which is part of the city's revitalization strategy for these central neighbourhoods, has these specific objectives:

- to increase the number of owners, and particularly resident-landlords, and
- to promote and facilitate sustainable homeownership for modest-income households.

The Domi-cible program has both a grant component and an education and training program component.

A third component, a renovation subsidy provided through the separate Rénove Atout program, was initially part of this program but is no longer available because it was oversubscribed.

The grant ranges from \$3,000 to \$8,000, and is available for purchasing a plex in the targeted neighbourhoods. The cost of grant is shared by the provincial government and the city at a 70:30 ratio respectively. The grant is based on creating greater parity between purchasing a single-family house and a plex. The amount of the grant was calculated to reduce the downpayment to roughly 5% for the average sales price in these neighbourhoods in 1999.

To qualify, the purchaser must agree to live in the building for at least three years. The building must meet these criteria:

- it must be exclusively residential and have two-five dwellings only;
- it must have been built before January 1991; and
- the value of land and building must not exceed certain limits (ranging from \$135,000 for a 2-dwelling unit to \$205,000 for a 5-dwelling unit).

To assist in sustaining ownership, a five-course program is being provided by two community-based groups. The courses address various pre-purchase (household budgeting, purchase financing and the purchase process) and post-purchase (property management and maintenance, landlord-tenant relations, and available subsidies) issues. In addition, personal counselling is also being offered [Wexler, 2002].

This particular program is part of wider efforts to support ownership in these areas that includes renovation grants to owners that are more generous for owner-occupied dwellings; grants to developers providing new dwellings through either new construction or conversion of nonresidential buildings; and modest tax credits to the purchasers of new dwellings [Wexler, 2002].

3.4. Affordable Housing Agreements

Ten provinces and territories have signed affordable housing agreements with the federal government within the last year. Out of these, seven contain provisions specifically for new affordable homeownership programs. These include the agreements signed by Alberta, Saskatchewan, Ontario, Nunavut, the Northwest Territories, Manitoba and Nova Scotia. British Columbia, Quebec and Yukon also have signed agreements, but without reference specifically to homeownership.

These agreements have been signed under a framework finalized on 30 November 2001 by the federal, provincial and territorial governments. It includes these provisions:

- Provinces and territories have the primary responsibility for design and housing program delivery;
- Provinces and territories require flexible programs to address their needs;
- The initiatives must create affordable housing for low-to moderate-income households;
- Units funded will remain affordable for a minimum of 10 years; and
- Provinces and territories are required to match federal contributions.
- The federal government will provide capital grants up to a maximum of \$25,000 per unit on average overall to increase the supply of affordable housing.

Under these agreements, the funds will be used for the following programs related to affordable homeownership:

• In Saskatchewan, for a 'Home Ownership Plan' that will support new construction and major renovations for low- to moderate-income families;

- In Alberta, for an 'Affordable Housing Partnership' that will provide stable, affordable rental or ownership housing for low-income families, seniors and individual with special needs;
- In NWT, for an 'Independent Housing Program' that will enable low income families to access modest housing through homeownership;
- In Nanavut, for a 'Downpayment Assistance Program' that will help first-time buyers purchase or construct a new home;
- In Ontario, for 1) an affordable urban home ownership program that will assist rental households to purchase newly-built affordable homes in urban neighbourhood revitalization areas and 2) an affordable remote housing program to create or rehabilitate ownership or rental housing in remote areas;
- In Manitoba, for 1) a downpayment assistance program for first-time buyers of new or existing homes, and 2) a new homeownership supply program for neighbourhood revitalization and remote areas;
- In Nova Scotia, for a new home purchase program to facilitate homeownership for low-to-moderate income households through new construction or conversion.

Affordable housing with respect to homeownership means housing that sells at or below average market price in the community or area. The provinces and territories are responsible for defining any income limits for the programs.

In urban areas, only up to 25% of the total federal funding can be used for homeownership, and it can be used only in neighbourhood revitalization areas. In remote areas, there are no such restrictions for affordable homeownership programs.

4. STAKEHOLDERS' MEETINGS

The following is a summary of the main observations made at a series of stakeholders' meetings held in Saskatoon, Winnipeg, Toronto, Montreal and Halifax in May and June of 2002.

The purpose of these meetings is to obtain local and independent input on the appropriateness and potential viability of utilizing Individual Development Accounts (IDAs) to improve the access of lower-income individuals and families to affordable housing.

At each of the stakeholders' meetings, the discussion initially focused on the potential for using the IDA savings to provide assistance towards downpayments and other initial costs associated with first-time home purchase. Subsequently, the discussion was shifted to exploring other housing-related uses for the savings.

Approximately 15 stakeholders were selected for each of the meetings to represent a range of sectors including provincial and municipal governments, local foundations and charities, financial institutions, community-based organizations, housing agencies and others (see Appendix 1). The stakeholders were selected because they were involved in some capacity in the provision of affordable housing, and particularly in the provision of affordable homeownership whenever possible.

Do to the relatively new introduction of this concept to many of the participants, no attempt was made at this time to discuss the detailed design and operational aspects of the potential project/program.

In preparation for these meetings, an information package was sent in advance to all of the stakeholders.

4.1. Homeownership Assistance

Two overarching conclusions came out of the discussions in all of the cities. First, there was broad support for using government assistance to help lower-income households achieve affordable homeownership. Secondly, within that context, there was also was strong interest in exploring the use of IDAs as a way of providing this assistance. This support and interest came from all of the sectors involved in these meetings.

Affordable homeownership was supported because it was seen as providing many benefits. As identified in all or most of the meetings, affordable homeownership was particularly seen as providing the following:

- positive and valuable effects on the "quality of life', not only for the homeowners and their families, but also for the neighbourhoods and the community at large;
- another way of creating much-need affordable housing, and particularly for those not having access to social housing nor able to buy market-rate homes;
- a more cost-effective way of providing affordable housing than assisted rental housing because the amount of public investment was smaller while the dividends were greater; and
- an indirect way of helping the rental situation by freeing-up existing units at a time that no or little rental housing was being built.

Within this context, IDAs were considered to represent a particularly useful tool that potentially could supplement their existing efforts or perhaps operate as a stand-alone program.

Offsetting the support and interest just noted, there were these two notable caveats:

- In Toronto, the stakeholders were generally more cautious about the potential for effectively assisting homeownership, due in large part to the high cost of housing in the city.
- In many of the meetings concern was expressed that any assistance for affordable homeownership would reduce the assistance for badly-needed rental housing and for families in greater need. As one stakeholder noted, assisting homeownership may not be the best way to spend scarce government dollars for creating safe and secure housing.

The support and interest was strongest in Saskatoon, Winnipeg and Montreal. It is notable that these three cities already have been actively engaged in providing affordable housing, and also ownership housing for lower-income households. As noted earlier, the affordable homeownership programs in these cities include the following:

In Winnipeg, there is the Housing Opportunity Partnership (HOP), Rent-to-Own Program and Winnipeg Housing IDA Program.

In Montreal, the city currently operates the Domi-cible, and has previously assisted home ownership through property tax rebates, and in specific mixed-income redevelopment areas.

In Saskatoon, the city takes advantage of the province's Neighbourhood Home Ownership Plan (NHOP), and provides other assistance through its Housing Reserve Fund, the Saskatoon Housing Initiatives Partnership and the Affordable New Home Development Foundation.

Saskatoon, which has been successfully promoting homeownership in their inner-city neighbourhoods, is beginning also to develop a wider range of homeownership initiatives. The objective is to make homeownership accessible to all working households in the community, and possibly even those on welfare.

In Halifax, there are no affordable ownership programs currently active. Consideration of affordable homeownership appears to have emerged only relatively recently. Nevertheless, various precedents were identified, including the federal Rural and Native Housing Program, the province's "Little House Program" (also know as "Hearth Homes"), and an earlier co-operative self-build program.

As an indication of the potential support, it was noted that the Quebec government had recently proposed using some of the funding made available through the new federal affordable housing program for providing as much as \$45,000 per unit towards affordable homeownership. This support was eliminated however, due to an effective lobbying effort of social housing advocates.

While expressing an interest in affordable homeownership, many of the stakeholders in Montreal were also interested in the potential for using an IDA-type approach to enable tenants in co-op housing, and even possibly social housing, to accumulate personal savings.

4.1.1. Target Population

The stakeholders recognized that an IDA homeownership program on its own would be capable only of generating relatively limited assistance that could be used toward a downpayment and the associated initial costs of home purchase. As a consequence, it would be of use only to those households having sufficient income to qualify for a conventional mortgage.

The preliminary figures,⁶ as shown on the following table, were presented and reviewed (but subsequently revised to reflect the input of the stakeholders). The figures were generally considered to be a useful starting point for identifying the potential minimum income and downpayment requirements. Among other things, it clarified the wide disparity in requirements caused by wide house price disparity among the various cities.

The main revision made to these figures related to the house prices. The figures were revised upwards by \$10-\$20,000 because the lower figure was thought likely to include units needing additional costly improvements before they could be occupied.

It was recognized that the house prices are for resale units, and not new homes. The stakeholders noted that condominiums might be available at a lower price in some markets, but also recognized that any cost advantage may possibly be mitigated by the associated condo/maintenance fees.

There was general support for allowing the IDA savings to be used for the other transactions such as legal and moving costs associated with home purchase in addition to the downpayments. One stakeholder also recommended that certain incidental costs such as the purchase of essential appliances should also be permitted.

In light of these figures, strong interest was expressed in Montreal, Saskatoon and Winnipeg for combining an IDA program with local resources to make homeownership affordable to households lower than those indicated.

In Toronto, some stakeholders felt that targeting households earning over \$50,000 or more would not have political and public support. It was thought that any affordable homeownership programs would have to serve those earning \$40,000 or less, but there were no indications at the meeting that local resources could be marshaled to achieve this. It was also noted that a household earning over \$50,000 should be able to save effectively through a RRSP Home Buyers' Plan.

This was recognizably a simplistic preliminary assessment, but it did provide an adequate basis for these early discussions. A more comprehensive analysis would need to include at the very least the price of smaller and larger units suitable for other households, other factors affecting the potential loans (like the current interest rates and household debt load) and the provincial variations in legal costs and transaction charges.

In this preliminary assessment, the lowest-priced homes were based on units identified in the MLS listings in each of these cities in March of 2002. For reasons of convenience, the assessment pertains to closings only on two-bedroom homes. The minimum household income needed to afford this particular housing was calculated, assuming a minimum 5% downpayment and a borrowing ratio of 2½:1 for house price to household income. In addition to the minimum downpayment, an additional allowance, based upon 50% of the downpayment, was included to cover other initial costs associated with home purchase.

Initial Affordability Analysis

	Winni- peg	Sas- katoon	Tor- onto	Mon- treal	Hali- fax
Min House Price*	\$70,000	70,000	140,000	110,000	70,000
Mortgage Loan	\$66,500	66,500	133,000	104,500	66,500
Min Household Income:	\$26,600	26,600	53,200	41,800	26,600
Downpayment (5%) Other Closing Costs Total Up-Front Costs	\$ 3,500 \$ <u>1,750</u> \$ 5,250	3,500 <u>1,750</u> 5,250	7,000 <u>3,500</u> 10,500	5,500 2,750 8,200	3,500 <u>1,750</u> 5,250
Average Monthly Savings**	\$ 35	35	70	55	35

^{*} lowest-priced two-bedroom houses on market (MLS, March 2002)

NOTE: For reasons of simplicity in this initial calculation, the minimum household income needed to afford this particular housing was calculated assuming a minimum 5% downpayment and a borrowing ration of 2.5:1 for the house price to household income. These figures were revised for the final main report using more precise affordability calculations.

4.1.2. Eligibility Criteria

There was general support for setting eligibility limits to ensure that the assistance did not go to households that would be capable qualifying for a mortgage without the assistance.

It was recognized that the eligibility criteria must reflect the house prices in the individual cities where they were applied. Various stakeholders indicated that this country lacked suitable definitions and standards for measuring housing affordability and eligibility in homeownership programs and that the Core need income threshold(CNIT)/, Housing Income Level (HIL) and LICO thresholds specifically were not appropriate for this purpose.

Some stakeholders cautioned that the income eligibility thresholds should not be defined too tightly. Many of the households potentially benefiting from this program could be already carrying a high debt load. Therefore, to qualify for a mortgage, they effectively would need a higher household income than otherwise required.

Some stakeholders also noted that the eligibility criteria should not be based solely on one income. Assets should be considered and adjustments must be made for household size.

^{**} based upon a three-year period and 3:1 match.

4.1.3. Subsidy Protection

There was a mixed response to the need for protecting the public (and, possibly, even the charitable and other local) contributions to the matching deposits.

Some stakeholders, generally in Halifax, were concerned that the public funds used in this way would be private gain. It was indicated that these programs would not be acceptable to the public or the government, unless the funds were eventually recovered in some way.

Others, generally in Saskatoon, Winnipeg and Montreal were not or less concerned about this issue. In certain cases, this response was most likely due to the assumption that the public money would be relatively modest and/or the benefits would clearly outweigh the costs.

Many of the stakeholders were looking to use the IDA program as part of municipal revitalization efforts. Under these circumstances, the public investment was seen as readily defensible due to the perceived public benefits. Furthermore, it was considered that homeowners buying in these areas were themselves taking a financial risk, and were unlikely to make a windfall profit for the foreseeable future.

Some stakeholders also noted that providing new homebuyers with this sort of assistance was not much different than other widely accepted practices, such as providing developers with a subsidy for the development of affordable rental housing, or providing individuals with economic aid to start new small businesses.

If safeguards were included to recover the money at some time, some stakeholders cautioned that it must be done in a way that would allow the homeowner to acquire and retain a reasonable equity stake. Reference was made to the second mortgage used by Habitat for Humanity and the "share of the gain" clause used in the Rural and Native Housing Program.

Some stakeholders were more concerned about securing the long-term affordability of the housing than recovering the public investment. Securing the affordability of the housing was considered to be a critical objective that was important but difficult to achieve.

It was indicated that local contributors to these programs, such as the charitable foundations, could possibly require similar protections before they would participate in these programs.

4.1.4. Public and Political Support

According to the stakeholders, government assistance for homeownership was more likely to be supported by the public and politicians when certain criteria were met. Among the criteria that were identified at the different meetings, and not necessarily universally supported, were these:

- The assistance should be carefully targeted to those with a valid and demonstrated need.
- The assistance in the case of new construction should focus on modestly sized and "no-frills" housing.
- The homebuyers should have personal financial equity in the housing.
- The assistance should target households needing a "small boost" or a shallow subsidy.
- The assistance should focus on areas needing revitalization.
- The assistance should be recoverable, or used only for housing that remains affordable for a long period.

The one notable contradiction in these comments related to the subsidy amount. Principally in Halifax, but also to some extent in the other cities, providing only a shallow subsidy was seen as critical to garnering public and political support. In Montreal and Saskatoon, there was an indication that there would be support for deeper subsidies to help households in greater need. In Toronto, there were indications that a shallow subsidy would not be acceptable because it would help only households with relatively high incomes, but there were no indications that deeper subsidies could be made available to reach those with lower incomes.

As indicated by their experience, assistance for affordable homeownership used as part of revitalization efforts was considered to receive public and politician support most readily. Many stakeholders noted that the community and family benefits were generally perceived to clearly outweigh the public costs associated with these efforts.

One aspect of an IDA program that was particularly important to many of the stakeholders was that the recipients of the matching funds first had to make a commitment through their own savings. These stakeholders felt that commitment of this nature (or through sweat equity or other means) was necessary to ensure that the property was properly maintained. They referred to earlier homeownership programs that failed because they "gave" housing without any such commitment.

Various suggestions were made about how best to sell the program. Some stakeholders focused on the homeownership aspects. They felt that the public should be educated about the benefits of homeownership and informed of the various safeguards used in the program, such as the criteria used in selecting the participants and the commitment that must be made by them.

Others focused on the asset-building aspects, and saw this program and others like it as being part of a comprehensive approach to alleviating poverty. To be successful, the concept had to be promoted in terms of its "global" benefits, such as its impact on public health, job creation and community revitalization.

One stakeholder (who supported affordable homeownership program) noted it is important not to argue that ownership is better than rental housing. It should be promoted as a way of providing more options in housing, and not for replacing rental programs.

4.1.5. Education and Counselling Programs

The importance of both pre-purchase and on-going post-purchase education and counselling or "mentorship", as it was called in one meeting, for lower-income households involved in homeownership was widely recognized and strongly supported.

Education and counselling for lower-income homebuyers was seen as a key to ensuring that ownership was sustained over the long term. These homebuyers were often vulnerable to unexpected problems and changing circumstances because they were living on a tight budget. Some stakeholders noted that many past programs supported home purchase but did not help the buyers when facing post-purchase difficulties. Programs that provided subsidies for lower-income homebuyers but no education and on-going support were considered likely to fail.

The stakeholders variously noted that the education and counselling should be directed at ensuring that the homebuyers met the following:

- understood "what they are getting into" when buying a home;
- were aware of the financial commitment and discipline involved in home purchase;
- appreciated the problems associated with maintenance of a home; and
- understood all of the ongoing costs of homeownership; and were able to cope with unexpected financial difficulties.

One stakeholder noted that the programs should help the potential homebuyers to "self-select" if they wanted to participate. This was considered likely to get people that have a strong commitment, and willing to change their behaviour to achieve the necessary savings.

Another felt that mortgage qualifications of the participants and particularly, those already with a high debt load, should be assessed early in the process, so that the participants were warned of, and possibly helped to deal with, any associated problems.

4.1.6. Other Concerns

Various concerns were raised at the different meetings about affordable housing programs generally, and an IDA homeownership program specifically. Some of these are mentioned elsewhere in this report, but are also included here.

Regarding affordable homeownership programs generally, the stakeholders noted the following major concerns:

- Lower-income homeowners participating in these programs could be living within tight
 budgets and under a heavy debt load. This would leave them ill-prepared to handle any
 unexpected costs, such as those associated with unpredictable renovation problems and
 rising interest rates.
- Assisting homeownership could take government dollars better spent on providing affordable rental housing or helping those in greater need.
- Assisting homeownership could involve giving away public money from private gain.

Regarding an IDA program devoted to homeownership, they noted these concerns:

- The program must have sufficient flexibility to meet local objectives and needs, and to work together with local resources and programs.
- The program must be capable of addressing two major and related problems: the serious difficulty that lower-income people have in saving, and also their high degree of indebtedness.
- Based on experience with *learn*\$ave, national demonstration projects appear to impose various research and monitoring requirements that are restrictive and costly.
- The program particularly, the homebuyer education and counselling will be expensive to operate. These costs might be too high, especially in smaller communities where the number of households helped might be limited.
- There are special challenges with aboriginal peoples related to treaty rights, their tax treatments, and possibly individual concerns about wanting to deal with only their own agencies.
- The administration of these accounts could be burdensome for the lending institution, and costly relative to any profit that might be derived.
- The tax status of the matched contribution must be resolved so that the benefits of the matched contributions are not taxed away.
- Similarly, the tax status of charitable contributions made to such a program must be
 resolved because all of the participants might not fall within the income groups
 considered eligible for tax-receipt assistance.
- In Toronto, and possibly other high-priced markets, the cost of housing could adversely affect the program. First of all, the high cost of ownership housing will significantly

increase both the income and downpayment requirements. At the same time, the high cost of rental housing will limit the savings potential of the participants.

4.2. Other Housing Options

There was general support for allowing the IDA savings to be applied toward other uses beyond the initial costs associated with home purchase, and also for other types of housing beyond homeownership. These additional uses were needed so that a wider range of households could benefit from the program, and so that participants unable to achieve homeownership might still be able to take advantage of their savings.

The discussion was focused on housing-related options because a *Home***\$ave** was seen as essentially a housing program. Nevertheless, many of the stakeholders felt that the program should offer the personal savings to be used for children's education, retirement and other non-housing purposes should the purchase of a house be unattainable.

4.2.1. Home Repairs

Allowing the IDA savings to be used for home repairs by new homebuyers was generally supported. This was seen as facilitating the purchase of older properties in need of repair and being particularly important because many lenders will want essential repairs made to older houses before providing loans. This was also seen as a way of supporting neighbourhood revitalization.

On the other hand, some questioned the practicality of this provision because of the limited amount of money made available through the IDA savings. That money would be sufficient for certain repairs, but not wholesale improvements. Also, in many cases, and particularly in high-priced markets like Toronto, the downpayment and other home-purchase costs could fully absorb the available resources.

Allowing the IDA savings to be used for home repairs by existing homeowners, on the other hand, was considered to be impractical and possibly inappropriate. Although many existing homes need repairs, often the repairs need immediate attention, which leaves the owners without sufficient time to save. Also, one of the major groups needing this assistance are seniors; although they might merit help, providing assistance through an IDA program did seem to be consistent with its fundamental purpose.

For these various reasons, it was suggested that some other renovation program might be a more effective option for houses needing extensive improvements, located in high-priced markets and/or upgraded by existing owners.

4.2.2. Equity Co-ops

There was support for allowing the IDA savings to be used toward purchasing shares in a equity co-op, but it was considered likely to have limited application due to how this type of housing is typically financed. Due to the difficulty of raising mortgages for equity co-ops, they are typically

financed through personal loans that require a substantial equity stake by the borrower as security. This means there are relatively few equity co-ops for which the IDA savings could be used. In addition, the purchasers must provide an initial equity stake that is much larger than the typical downpayment saved through an IDA program.

It was noted that another emerging form of tenure called "co-ownership", which blends certain characteristics of condominiums and equity-coops, might be suitable for support.

4.2.3. New Construction

Using the savings for newly built homes was considered to be a reasonable option that should be left open, although the higher new home prices in most markets would stretch the required savings. This option was thought to merit consideration because new homes, which require less initial upkeep than older ones, could be better for households with limited incomes. This option could be made more practical where additional subsidies were available to lower the price of housing.

4.2.4. Rental Housing

Using IDAs to help people in shelters or tenants in inadequate private rental housing was put forward as a potential option. It was specifically asked if an IDA program could be used to help them gain access to permanent or more adequate rental housing by saving for the initial costs like the first and last month's rent (or security deposits), utility deposits and/or essential appliances.

While there was interest in assisting tenants to upgrade their accommodation, the stakeholders generally questioned the effectiveness and appropriateness of using IDAs for this purpose. Their doubts seemed to revolve around these two points:

- Most of these people would have limited capacity to save. Therefore, they would not be
 able to accumulate sufficient money, unless they stayed in their present inadequate
 accommodation for a relatively long time.
- Using IDA savings for rent did not appear to be consistent with asset-building because it did provide tenants with a tangible and permanent asset. The assistance would be depleted through the payments.

Reference was also made to other existing programs like the Calgary's security deposit program and Toronto's rent bank that were available for helping tenants in crisis. These were considered to be more effective option because they money up-front when most needed through loans, while requiring repayment at a later date.

It was recognized however, that many of these tenants could no doubt benefit from the financial management training and other supports provided through the IDA program..

The stakeholders generally were in support of programs that would allow renters to build an asset. One option preferred by many was a "rent-to-own" program. Another, as discussed in

the Montreal meeting, was a program that would enable co-op residents, and possibly even tenants in government-assisted housing, to build equity by voluntary savings made through their rent payments.

They also noted that many tenants in market units pay more in rent than they would pay in mortgage payments for similar accommodation. An IDA program would be very useful if it enabled this particular group of renters to purchase homes.

4.2.5. Non-Profit Co-ops

In Montreal, strong interest was expressed in using an IDA program to assist people living in non-profit co-op housing to acquire assets. They should be capable of saving because their housing costs are controlled. In addition, because they pay low taxes, they generally cannot take advantage of fiscal measures like tax credits. Such a program could utilize a central fund modelled on the FTQ (Fédération des travailleurs du Québec) fund, which is a labour investment fund into which workers make voluntary but incentive-based donations through automatic withdrawals from their paycheques. The fund could be administered by a financial institution like Desjardins, and the money in the interim used to finance co-operative housing development.

4.2.6. Secondary Apartments

Using the IDAs for creating secondary (or accessory) apartments was not generally supported. Many of these cities did not permit accessory apartments, except perhaps in limited areas. Some stakeholders considered this activity to be inappropriate for their revitalization areas, where it was likely to increase problems associated with too much and poorly maintained rental housing.

Many stakeholders also questioned whether this option would be practical. Creating the second unit by itself is a difficult, expensive and time-consuming task. The new homeowner must be capable of undertaking or managing this work, and then becoming a landlord, while servicing the mortgage on what presumably is a tight budget. All of this was considered likely to over-stretch the capabilities of most new homeowners.

Against these concerns, some stakeholders noted that secondary apartments historically have been used by lower-income individuals and families as a means to afford homeownership. Furthermore, these units typically provide much-needed inexpensive rental housing.

4.3. Local Participation

The operation of any local IDA program, it was widely appreciated, would depend upon the participation of locally based community organizations. To various degrees in all of these cities, there appears to exist a network of local organizations both capable of, and interested in, supporting and operating an IDA program. These established networks include community-based organizations, local foundations and charities, lending institutions, local governments and other groups.

As previously noted, support was expressed in Saskatoon, Winnipeg and Montreal for using local resources to make homeownership more accessible for lower-income households, including those with relatively limited means. This probably would involve engaging and integrating various tools, organizations, and subsidies.

It was frequently noted that to succeed any IDA program should allow sufficient flexibility so that it could be "customized" to meet local needs and priorities. Various possible local preferences were indicated. Winnipeg and Montreal appear to be most interested in focusing affordable housing assistance in inner-city neighbourhoods. Halifax indicated an interest in supporting the provision of new modestly sized units, and possibly pre-fabricated units. Montreal was also interested in developing a program specifically for existing co-op residents.

Concern was expressed about the cost of operating these programs generally, and the education and counselling component specifically. Some of these costs, it was appreciated, were related to start-up and research. Nevertheless, the community-based organizations indicated that are likely to need operating assistance.

The local foundations and charities indicated their interest in supporting IDA programs. As noted by one stakeholder from this sector, but also reflected in the comments of others, these

organizations should be involved in programs improving the quality of life in the city generally, and not just serving the "poorest of the poor".

The local credit unions generally appear to be open to participating in an IDA program. At least one had already looked into starting financial management training, and using more flexible mortgage underwriting criteria. Nevertheless, some of the representatives expressed concern about the potential burden and cost of administering the many small accounts. They indicated that they would want help from the community-based social agencies to process and "handhold" the participants.

It was indicated that the local matching funds for the IDAs could possibly be made available through the new federal supply program, but might depend upon the individual provincial agreements.

5. RENT BANKS

Rent banks are recently emergent programs that essentially provide modest loans for tenants seeking to secure an apartment and/or to retain an apartment in face of potential eviction. This relatively low-cost form of assistance is seen as an effective way of reducing the pressure on shelter housing and the need for other more costly assistance.

Rent banks are now widely used across the US, although the first documented rent bank, Connecticut's Eviction Rent Bank Program, was established only in 1989.

The rent banks in Canada are more recent. The first program was a twelve-month pilot project in Toronto was started in September 1998 in response to declining vacancy rates and skyrocketing rents. Since then, Toronto has proceeded to establish an on-going rent bank as well as a separate Shelter Fund for families on social assistance.

In the short intervening period, a large number of Ontario communities have followed this precedent. As of late 2001, there were rent banks in Belleville, Brantford, Durham, Hamilton, Muskoka, Orangeville, Ottawa and Woodstock; and at least another being organized in Kitchener-Waterloo [Wingard, 2001].

No attempt has been made in this study to identify all of the rent banks across the country, but enquiries were made with knowledgeable people across the country contacted in the course of this study. So far, similar programs were found only in Calgary. Enquiries will continue during the remainder of the study.

This report examines the experience in Toronto, Hamilton, Ottawa and Calgary, identifying their different objectives, practices and results, and some of issues being raised.

5.1. Toronto: Rent Bank (Pilot Program)

Toronto's pilot rent bank program was introduced in September 1998 as part of an eviction prevention strategy. The twelve-month project was established "to determine to what extent a small loan and/or grant can keep a family housed, thus diverting them from the shelter system." It has been the subject of a published evaluation [LaPointe, 2001].

Under this program, loans were provided mainly for mother-led, single-parent, tenant households renting in the private sector, and who were usually able to support themselves but were facing eviction due to some short-term problem. This group was targeted because they were considered to face the greatest likelihood and also the worst consequences of eviction.

A total of \$80,000 in funding was supplied by the following:

- \$55,000 from the city's Homeless Initiatives Fund, of which \$40,000 was set aside for loans;
- \$5,000 from United Way for administration;
- \$10,000 from CMHC for evaluation of the project; and
- \$10,000 for loans from Hockey for Homelessness, a charitable organization.

The program was operated by four downtown non-profit organizations: Neighbourhood Information Post, Regent Park Community Health Centre, Central Neighbourhood House and Dixon Hall. These organizations provided the staff, offices and other resources. The staff time amounted to an additional estimated contribution of \$65,650 for the first six months alone.

In April 1999, before the conclusion of the pilot period, the city established its Shelter Fund, which is described later. As a result, some of the rent bank's clients changed because they became eligible for assistance from this other source. The rent bank negotiated with the Shelter Fund for a repayment of \$16,800 in outstanding loans when they were transferred.

5.1.1. Results

The program provided loans to 39 families. The loans ranged from \$250 to \$2400, and averaged \$1100. Most loans were to pay arrears, but in a few case small amounts were also provided for utilities.

The household incomes ranged from less than \$10,000 to over \$20,000, while the average was \$14,700. 45% of them paying 70% or more of their income on housing.

Their average rent was \$830, mostly for two-bedroom units. This was close to the average of \$880 for a two-bedroom unit at that time.

Most of the recipients' housing situations had improved or stabilized six months after the loans had been provided. The loan helped 65% of the clients to remain in their homes, while another 6% were able to move to better housing. Only one client was known to move into an emergency shelter.

As of April 1999, less than a third had made repayment plans, and only two had kept to their repayment schedule.

The administrative costs for the pilot were high — partly due to the start-up costs and to the high-need tenants.

A cost assessment indicated that the savings to the city of this program amounted to \$65,000-389,000 annually. This was based on the assumption that 25-50% of the clients would have used city-funded shelters without this assistance.

5.1.2. Recommendations

The final report contained many recommendations, including the following:

- The rent bank should continue to target the loans primarily to the following tenants:
 - o those facing one-time and short-term financial emergency, such as those caused by sudden loss of income, health problems, or an accident.
 - o those able to repay the loan, while recognizing that all or part of it might not be repaid in some circumstances.
 - o those renting market rents in the private market or in social housing projects.

 Only in an extreme emergency should assistance be provided to rent-geared-to-income tenants.
- The rent bank should not be used to assist those eligible for the Shelter Fund, but the
 funds could be used in this way in an emergency and then recovered from the Shelter
 Fund.
- The rent bank should not be restricted to families, but higher priority might be given to them.
- The city should remain responsible for monitoring how the funds are spent, while the community-based agencies should be responsible for the administration and delivery of the services.
- The rent bank services should be handled by existing organizations providing similar services, in order to allow for "one stop" shopping and to reduce staff costs.
- The rent bank staff should undertake the following:
 - intake and approval of loans;
 - o assistance with personal budgeting;
 - o negotiations with landlords;
 - liaison with other relevant agencies;
 - o information and referrals; and
 - follow-up on loan payments.

5.2. Toronto: Rent Bank ("Expanded" Program)

Toronto started its "expanded" rent bank in December 1999. The initial funding of \$200,000 came from the city's Homeless Initiative Fund, which in part was supported by the Provincial Homeless Initiatives Fund. Since then, the city has annually renewed this funding (with a slight

increase for inflation) for two additional years. Supplementary funding was also provided through \$25,000 from United Way and two private amounting to \$10,000.

This program was established following a recommendation in January 1999 of the Mayor's Homeless Action Task Force, which called for creating a \$500,000 rent bank for tenants at risk of eviction due to arrears.

The rate of evictions from both private and public sector housing in the city had increased steadily over the 1990s. The landlord-tenant legislation had not been effective in protecting low-income tenants from high rent increases. Almost 40% of the families seeking emergency shelters had been evicted, and 75-85% of evictions are due to rent arrears. Impending changes, enacted to Ontario's Tenant Protections Act 1998, was expected to make matters worse [Hoy, 1999].

The purpose of the program is to sustain current tenancies by preventing evictions whenever possible. It was premised on findings from Toronto and cities in the US that it is much cheaper to prevent someone from becoming homeless than it is to accommodate them in emergency shelter or motels. It also has various social benefits, including preserving the stability of the lives of families and their children.

The rent bank program is administered by the Neighbourhood Information Post (NIP), which is responsible for disbursing the funds, monitoring the results and reporting to the city. The program is operated through seven outreach centres, including NIP and six other existing housing help centres or similar community-based organizations. These organizations were selected by the city, on the basis of proposals, because they provided good geographical distribution across the city and had had previous experience in offering related housing services.

In addition to operating the rent bank, these centres also provide the following associated assistance:

- mediation with landlords or property managers, for example, to work out a rent payment plan;
- assistance in obtaining Community Start-Up Benefit and City's Shelter Fund for those receiving or eligible for social assistance;
- counselling and advice, particularly to help prevent clients falling into rent arrears;
- information on the Tenant Protection Act;
- referrals to other social service agencies;
- assistance in finding alternative rental housing; and
- advocacy with government agencies.

5.2.1. Conditions

Under this program, loans are made available to families with dependent children living in private rental housing that have received a formal eviction notice. This represents a widening of the eligibility used in the pilot program, which targeted single mothers.

The assistance is provided solely for rent arrears. Those having utility arrears are directed to another local program — *Share the Warmth*.

There is no formal limit to income eligibility. The clients are expected to have exhausted other means, including the possibility of a bank loan.

The assistance is not available for those receiving or eligible for social assistance (which is handled through The Community Start Up Benefit or Shelter Fund), or paying an unsustainable proportion of their income on housing — usually taken to be 70%.

5.2.2. Results

During Dec 1999 through Dec 2001, the program assisted 415 families (or 1,369 people in total). It provided loans amounting to slightly over \$599,000, or nearly \$1,445 per family. Over \$139,100 had been repaid in that time, or about 23%.

The average income for these families was about \$1,950 per month or about \$23,500 per year. Their average monthly rent was about \$825 in the first year, and \$905 in the second — representing an increase of nearly 10%. On average, the families are paying about 45% of their income for rent.

There has been a steady increase in loan defaults over the life this program, due in large part to rental increases and stagnant wages.

Figures from the Dec 1999-Jan 2001 period indicated that the loans prevented eviction in 87% of the cases.

5.3. Toronto: Shelter Fund

This fund, which was established in April 1999, provides assistance for families with dependent children living in Toronto that receive social assistance through Ontario Works (OW), Family Benefits or Ontario Disability Support Program (ODSP), or are eligible for this social assistance.

The fund was designed specifically to assist the following:

- households without permanent shelter to obtain suitable housing;
- households with suitable housing to maintain it, particularly in the face of possible eviction; and
- households in unsuitable housing to secure more suitable (including more affordable) housing.

The fund was created using a \$6 million saving that resulted from the introduction by the federal Canada Child Tax Benefits in July 1998. As part of this change, the province made a commitment to reinvest the social assistance savings into programs that benefit low income families with children, and also to encourage the municipalities to use their savings in a similar

manner. This particular approach was taken because the majority of social assistance families in the city pay more than the maximum amount they receive under OW for accommodation.

This assistance is generally provided through one-time payment, except for the two types of transitional assistance provided on a monthly basis as described next.

Subsequently, the program was expanded to provide special funding for the following specific families with dependent children:

- under the so-called "shelter enhancement", families who are actively participating in an approved OW activity and are paying shelter costs in excess of the OW legislated shelter allowance. This assistance is paid monthly for a maximum of six months, but extensions are possible.
- under its "shelter enhancement for hostel integration support", families who are not necessarily involved in OW activities but who upon leaving the hostel system have found accommodation at a cost exceeding the OW legislated allowance. This assistance is paid monthly for up to three months, but extensions are available dependent upon the client participating in an approved OW activity.

The assistance is available for the following families:

- those moving into rental or boarding accommodation that is not "per diem funded", like group homes or hostels;
- those being discharged from an institution or other approved facility;
- those leaving a harmful situation; and
- those moving to less expensive accommodation or for reason of employment, education or training.

Harmful situations include a variety of conditions: fire and flood; chronic homelessness; family violence, including spousal or child abuse; uninhabitable or unaffordable premises and others.

The assistance is available for these direct shelter costs:

- last month's rent deposit;
- fuel and hydro deposits for new services;
- rental, utility or fuel arrears; and
- moving and storage costs.

The Community Start Up Benefit is available for other second shelter support costs, including household furnishings such as beds, dresser, table and chairs; and settlement costs such as bedding dishes, small appliances and clothing.

A service plan must be developed with each client, in order to assist each in managing shelter costs.

The support available under the two enhancements is based upon the difference between the maximum OW shelter allowance and the actual shelter costs, up to the following limits:

- \$145 per month for a family of 2;
- \$183 for a family of 3-4; and
- \$285 for a family of 5 or more.

5.4. Ottawa: HomeSafe

HomeSafe is a rent bank program started in February 2000 and operated by the Community and Family Services of the Salvation Army in Ottawa. It is funded by the City of Ottawa through the Provincial Homelessness Initiative and the federal SCPI program.

Salvation Army had been providing a similar service before this funding, but on a more limited and ad hoc basis and without a committed budget and stated eligibility guidelines.

The program provides loans for families with children who are at risk of losing their homes because of payment arrears for rent or essential utilities. This service was developed to prevent families from turning to shelters for emergency housing.

The loans are available only to these families meeting these conditions:

- living in housing that is permanent and habitable
- not owing more than 2 months in rent arrears, or more than 3 months of utility arrears or other fees;
- not involved in chronic financial mismanagement;
- able to repay the loan in a timely manner;
- having exhausted other social services and community resources; and
- not having used this program previously.

There are no income limits to eligibility. Applicants are reviewed on case-by-case basis. The key criteria is that applicants have exhausted all other resources, including their own.

All clients are requested to sign a repayment plan based upon their income. No families are denied assistance if they are unlikely to be able to repay. The recipients are still asked to sign because most would rather see the assistance as a loan than a hand-out.

5.4.1. Results

In 2001, the organization received approximately 820 enquiries about getting assistance. It assisted 240 families, referred 270 to other agencies or supports in the community, and denied assistance to about 310 — either because they did not meet the criteria or were helped by other agencies.

The assistance typically was about \$1000-1100 per family for rent, and \$300-500 for utilities.

Only 14% of the recipients have made repayments. This figure includes both complete and partial repayments.

Most of those assisted are families led by single mothers not receiving sufficient child support. Some are working with top-ups from social services; others are solely dependent on social services.

They also helped many low-income double-income families with children, earning about \$30,000 through minimum wage jobs with irregular working hours. They would be eligible for social housing if any was available.

Many of the families are paying 60% or more of their income to housing.

Under the earlier ad hoc program, from April 1999 to November 1999, they provided \$31,250 in loans to 49 families at an average of nearly \$640. About \$1,620 had been paid back by the end of that period.

5.4.2. Other Services

Salvation Army also assists families and singles not just those getting financial assistance in a variety of ways:

- advocating and negotiating with landlords, utility companies and other community service providers;
- providing information on tenant rights and obligations;
- providing information on other community resources offering assistance and support; and
- providing other assistance in the form of groceries, furniture and clothing.

The organization also has recently started a pilot project, called *MoneyWise*, and funded by Ottawa through the Provincial Homeless Initiative. It is directed at decreasing the risk of homelessness through early intervention into the financial management of the vulnerable families. It is intended to keep families in appropriate housing by stabilizing their financial situation before they are threatened by eviction.

The project provides a financial counselling and also a trusteeship service. The organization will develop individual budget and payment plans, based on the income of the client. Where necessary or appropriate, as part of those plans, it will also take over the disbursement of spending money. One of the intents is to secure affordable housing for the clients by assuring that their rent will be paid on time.

5.5. Hamilton: Housing Emergency Loan Program (HELP)

HELP was started in December 1999. It temporarily ceased operation in May 2000 when its funds were exhausted, but re-opened in December 2000.

It was founded by a number of faith-based groups and social service agencies, which included the Unitarian Church, Downtown Ecumenical Committee, Social Planning and Research Council, McQueston Legal Services, Dundurn Legal Services, Housing Help Centre and the Solutions for Housing Action Committee.

HELP is a stand-alone and non-incorporated organization program devoted exclusively to administering the loan program. It provides no other support services, and employs one full-time staff person. All funds are administered through the Hamilton's Housing Help Centre. Public donors are issued charitable tax receipts through the United Way. It is governed by a Steering Committee comprised of representative from the founding organizations.

The initial funding came through \$10,000 from the Unitarian Universalist Fund in the US, and \$32,000 from the Regional Municipality of Hamilton-Wentworth. A grant of \$120,000 was subsequently provided by the federal SCPI program. Other local funding has provided by Arriscraft International, Centenary United Church, Hamilton Community Foundation, St Thomas More Secondary School, and private donors.

The purpose of the program is to aid low-income tenants to secure rental accommodation, both subsidized and market. It is available to families, couples or individuals whether employed, or on pension or social assistance, having an income at or below 110% of the LICO.

The program provides no-interest loans that repayable over one year. The loans are for one-time costs in the form of moving expenses, first and last month's rent, and other associated costs like providing a major appliance. The loans are now limited to a maximum of two months' rent; until recently, this limit was one month.

The loans are given only under these conditions:

- The rent of the unit is sustainable that is, within the range of what the tenant can afford.
- The tenant intends to stay in the unit for at least one year.
- There is a reasonable expectation that the loan will be repaid within one year.
- The tenant is faced with a temporary not chronic crisis that can be addressed through one-time assistance.

5.5.1. Results

From Dec 1999 through May 2000, the program received over 200 applications, and gave 93 loans totaling \$46,000, or \$494 per loan on average.

From Dec 2000 through Aug 2001, it received 97 applications, and gave 74 loans totaling \$40,100, or \$542 per loan.

About 3/4s of recipients were making payments, either regularly or irregularly. Overall, less than 50% of the borrowed amount has been repaid according to the repayment agreements.

The recipients of the loans were 44% singles, 8% couples, 28% singles with children and 20% couples with children.

The majority of applicants had monthly incomes between \$500 and \$1,500. The average was \$1,225. For comparison, the LICO in 2001 for a couple in Hamilton was \$18,367 per year, or \$1,530 per month. The average rent for an one-bedroom apartment in Hamilton in October 2000 was \$525, while the average for a two-bedroom was \$695.

Half of its applicants are on ODSP. Applicants that rely on OW, on the other hand, are rarely eligible because they cannot meet repayment requirements.

5.5.2. Observations

Hamilton's program has somewhat different objectives than that in Toronto. The Toronto program is for families already served with an eviction notice. These families are often in deeper financial trouble, requiring more money and facing greater problems in paying it back. As a consequence, the Toronto program acts more like a grant program rather than revolving loan program. *HELP*, on the other hand, is aimed primarily at helping tenants secure rental housing. It also limits the amount of the money that it will loan.

Their requirement that the loan be repaid is disqualifying many people in great need. Included in these are many of the homeless and chronically ill, and those on OW.

The organization is giving consideration to associating with another organization. This move is seen as a way getting access to charitable access, serving clientele in other parts of the city, and making better use of their resources.

5.6. Calgary: Security Deposit Programs

At least four organizations in Calgary have recently run security deposit programs. These include Red Cross, Salvation Army, MCC Employment Development and Calgary Urban Projects Society (CUPS).

These programs respond to a very specific problem in Calgary. A study found that over half of the people seeking help in the emergency shelter system were working. For example, many of them were newcomers to the city, who had the income to afford a rental unit but did have the last month's rent as the security deposit.

These programs were designed to assist tenants in securing housing, but not to prevent loss of housing. They do not provide assistance towards other financial needs like utility deposits, nor rental and utility arrears.

Most of these organizations provide loans for security deposits. For example, MCC provides loans only to those that are employed. The recipients must take a money management course. The loans must be paid back over two years.

CUPS is the only one of these that provided grant. It targeted mainly those on social assistance. In 18 months, they gave out \$543,000 to nearly 1900 applicants, or an average grant of \$290. About 1/3 of the recipients had mental health problems; this group was the focus of most of their other services. The funding came from SCPI, the Calgary Homeless Foundation and United Way. This program has been suspended due to lack of money.

6. HOMEBUYER EDUCATION AND COUNSELLING PROGRAMS

Homebuyer education and counselling programs (HEC) are widely used across the US as a way of increasing homeownership in traditionally underserved markets, while at the same time limiting the mortgage loan default risk.

Those "underserved" by the conventional mortgage market are defined by their low homeownership rates. They variously include lower-income households as well as minorities, young households, women-led households, inner-city residents, recent immigrants, rural populations, aboriginal groups, single-women, and self-employed and others in non-traditional work arrangements.

Increasing homeownership in these markets is considered potentially risky by private lenders because they cannot meet conventional mortgage underwriting criteria. These underwriting conventions have been established by lenders to generate high lending volumes within tolerable default losses and acceptable administrative costs. They can create barriers, however, for many credit-worthy and nearly creditworthy customers that do not fit the standard profile. HEC finds creative ways of addressing many of those barriers.

HEC was the subject of a recent CMHC report [Hirshhorn, 2000]. The following provides some of the discussion within that report of the following, but not all, was taken from that report.

6.1. Background

The first "housing counselling" program was identified in federal legislation in 1968. Over the intervening years, the use of HEC has grown from a few pilot projects to well over a thousand of local providers across the country, a handful of non-profit umbrella organizations that provide training for the local agencies, and a national association representing all of the stakeholders. Most of that growth has occurred in just the last 10 years.

In the early years, HEC focused on post-purchase counselling as a way of preventing mortgage foreclosure. In the last ten years or so, pre-purchase education and counselling has been used as an instrument of public policy to extend homeownership. Many agencies now provide comprehensive "full-cycle" services designed to help families both before and after purchase.

Another recent and key development is the HEC providers taking on the role of an "intermediary" between the prospective homebuyer and lender. This sees some of them more aggressively facilitating home purchase by finding alternative ways of qualifying for mortgages, or even undertaking the underwriting functions on behalf of the lenders.

Various factors have contributed to the increased and wider use of HEC. Among the key factors are these:

- In 1974, HUD was authorized for the first time to fund HEC programs. Starting with regular annual appropriations of \$3 million to establish counselling agencies, its budget devoted to HEC has increased to \$12 million in the 1990s and \$18 million in 1997.
- The Community Reinvestment Act, passed in 1977 but not effectively enforced until 1989, provided a way for community groups and others to pressure lenders to increase their lending efforts in the underserved markets.
- During the early 1990s, many state housing finance agencies and local governments
 developed housing programs to assist first-time or lower-income homebuyers through
 such devices as downpayment grants, soft and silent second mortgages, and low-interest
 mortgages. Access to these products was generally contingent upon passing a certified
 HEC course.

- Since 1992, HUD has set affordable lending targets requiring Fannie Mae and Freddie
 Mac to perform a large proportion of their lending in underserved areas. (These
 powerful quasi-government agencies are secondary lenders; they do not underwrite
 mortgages directly, but they do purchase loans from originators that conform with their
 standards.)
- In 1994, Fannie Mae began requiring completion of certified HEC programs in order to obtain various affordable products or other housing assistance. In the same year, Fannie Mae was persuaded to accept tele-HEC to satisfy this requirement.
- In 1995, the "National Homeownership Strategy" provided additional support for various measures — including HEC — directed at increasing homeownership rates in the US.

The mortgage loan insurance provided through the Federal Housing Administration (FHA) provides another important support for HEC. Under this program, up-front rate reductions of 30-40% are provided for insurance premiums to first-time homebuyers completing accredited HEC programs.

A national association, the American Homeowner Education and Counseling Institute (AHECI), was established in 1996 to represent and co-ordinate the various stakeholders that have encouraged, funded and/or participated in the delivery of HEC. Among the primary objectives of this organization are establishing standard certification procedures, evaluation criteria and core curricula for these programs.

Also, as part of its unstated mandate, it is expected to determine how best to share the cost of these programs, which is an issue that divides the various parties involved.

6.2. Experience in the US

The HEC industry has developed in a haphazard fashion in response local needs and capabilities, and then has been molded by various financial and institutional considerations as indicated from the national level. This combination of "bottom-up" innovation and "top-down" support and guidance is reflective of how most of the new housing initiatives have emerged in the US for the last twenty or more years.

The US industry, as a result, has been highly fragmented. It has involves a variety of service providers, many of which have developed their own programs, procedures and standards. Only fairly recently, mainly through HUD and AHECI, has there been a move towards more consistency.

6.2.1. Stakeholders

HEC involves a mixed range of stakeholders. The principal ones are diverse non-profit organizations that mainly provide the services and private lending institutions that provide the

mortgages. Also involved are private credit management companies to some extent, and federal, state and local governmental agencies.

Locally-based non-profit organizations are the main providers of HEC. Some are independent, while others are affiliated with one of the national umbrella organizations — chiefly, Neighborhood Housing Services, ACORN, Neighborhood Reinvestment Corporation and Catholic Charities USA.

These agencies often were first established — and still continue — to be mainly involved in other activities, such as developing and rehabilitating affordable housing, managing non-profit rental properties, supporting community or economic development, and organizing and running community-based social programs.

According to one survey, the average non-profit agency involved in HEC had been in operation for more than 18 years, but has provided HEC for no more than 5 years. The average agency offered nine hours of classroom instruction and 2.5 hours of one-on-one instructions. The average agency provided assistance to nearly 500 households in the last year.

Many of the organizations, due to the nature of the community and the focus of the organization, have developed programs specific to certain underserved groups.

6.2.2. Clientele

The HEC providers are able to sort out and work with principally two types of clientele:

- The "creditworthy": those able to purchase a home, but not having done so for various cultural, social or linguistic reasons.
 - Included here, for example, are homebuyers unable to qualify for home purchase because they do not have a credit or banking history. For example, many immigrants operate solely on a cash basis. To establish their creditworthiness, HEC providers might use alternative "non-traditional" means like showing that the client has a history of timely paying rental or utility bills.
- The "near creditworthy": those not able to meet all of the underwriting criteria, but able do so with some assistance.

This includes those that have a marred, but reparable, credit history because of some past financial problem. HEC can assist by instituting repayments or settlements with those owed money, developing feasible budgets and promoting fiscal discipline, and ultimately providing clients with a favourable credit report.

6.2.3. Services

The service providers are variously involved in a range of education and counselling activities — both pre- and post-purchase. Education covers a wide range of subjects in a general way, and is mainly undertaken in groups in a classroom setting. Counselling is one-on-one and deals with specific personal problems. The latter is the more time-consuming and expensive exercise.

Pre-purchase education is generally directed at helping the households to decide whether home purchase is right for them, and to understand the home buying process. The courses typically cover topics like determining what housing is affordable, understanding housing finance and the closing process, preparing a household budget and establishing fiscal discipline, recognizing the importance of maintenance and repairs, and avoiding default and foreclosure.

Pre-purchase counselling is directed generally at screening clients to determine whether they are ready and able to purchase a home, and helping them overcome any financial, credit, linguistic, cultural or other barriers. It will typically include most or all of the following:

- assessing a family's ability to buy a home based on income and credit history;
- advising on the management of household expenditures;
- setting savings goals to finance downpayment and closing costs;
- helping to establish a credit history or repair a marred one;
- screening clients for home purchase assistance (like downpayment grants) offered by local governments or housing agencies;
- assisting clients in finding employment to increase and/or stabilize incomes; and/or
- linking clients with lenders, real estate agents and social service providers.

Post-purchase counselling focuses on overcoming any mortgage repayment problems, and particularly on preventing default and foreclosure. This might consist of assisting in money management, providing temporary financial assistance, making referrals to other agencies, and obtaining forbearance from the lenders.

Some agencies also offer post-purchase education programs that deal with such aspects as home maintenance and repairs, and alternative financial arrangements like life leases and reverse mortgages.

6.2.4. Involvement as "Intermediaries"

What is particularly notable about some of the HEC agencies is that is that they offer services aimed at facilitating access to homeownership. This involves actively "packaging" the prospective homebuyers as good customers. They do this by addressing past credit problems, finding alternative ways of qualifying them, and intervening on their behalf with the lenders. Sometimes, this is done through affiliated lenders who hand over the underwriting to them.

Due to this activity, some of organizations fall into an important new class of agencies in the US called "intermediaries". What chiefly characterizes these agencies is that focus on getting access to private capital for lower-income households, micro-businesses, and non-profit developers rather than on administering government-supplied assistance.

It is important to that this process does not typically involve the lenders relaxing underwriting criteria or providing concessionary interest rates. (Financial assistance — like downpayment grants or even low-interest loans — might be available, but only through governmental programs.) It does involve the lenders in accepting alternative ways of meeting their criteria. In

doing so, they are essentially accepting a risk assessment for these clients so that they have access to more favourable market rates available to other good customers.

6.2.5. Role of Lenders

The mortgage lenders have become supporters of HEC, mainly since the enforcement of the CRA. Their interest can be seen the emergence of pre-purchase homeownership education. This represents for them a low-cost marketing strategy and screening process that enables them to meet CRA compliance and HUD-imposed lending mandates.

HEC has emerged for these reasons as a key tool for the lenders to penetrate the underserved market. This is an important aspect of HEC that should not be overlooked, and explains much of its success. While the increased lending in these markets has been accomplished through new mortgage products and underwriting standards, these initiatives are often directed at households that face various problems with home purchase. Because the lenders have little information on these markets and little experience with alterative methods for evaluating lending risk and establishing creditworthiness, they typically find that identifying good customers in this market to be too expensive and time-consuming on their own.

The lenders have turned to the non-profit HEC agencies both to market their products and process the applicants. For their part, the agencies have developed successful outreach methods to the underserved communities. They have succeeded in making home purchase process less daunting for families, sorting through the underserved populations for creditworthy and nearly creditworthy households, helping them to address and repair any shortcomings in their mortgage qualifications, and ultimately sending the lenders "mortgage-ready" households. This has enabled the lenders to increase their lending volumes and make a profit.

6.2.6. Tele-HEC

Over-the-telephone HEC, called tele-HEC, was aggressively implemented by two large private companies in 1992. Some of the largest providers of pre-purchase education now use this approach. Tele-HEC companies are able to process large numbers; some deal with 200-300 clients per month. More than 25% of the HEC recipients received training over the telephone. In tele-HEC, the homebuyers learn from workbooks sent by the service provider. They are helped over the telephone with any questions, and also tested in this way to verify their command of the material.

Tele-HEC is considered to be limited type of pre-purchase education. The reason for its popularity is that it is the least costly way of qualifying for various government programs that require HEC.

Tele-HEC, on the other hand, is not considered to be an adequate way of providing counselling, and particularly it is not considered effective in limiting default risk. Notably, FHA does not recognize tele-HEC for its reduced mortgage insurance premiums.

6.2.7. Costs

A 1997 survey found the average cost of counselling and education services was \$650 per prospective homebuyer. The costs ranged from \$100 to \$3200, with high-volume education courses at the low end, and counselling and full-service at the top.

As indicated by figures provided earlier, HUD has been provided significant and on-going financial support from HEC. Over the years, HUD has funded about 25% of the costs. The remainder the costs are variously split between the private lenders, local governments, non-profit agencies, other contributors as well as the homebuyer.

Overall, governments on average have provided so far over a third of financial support. Lenders have provided about one-third of the pre-purchase costs, but notably much less for the post-purchase.

The information on charges to homeowners is inconsistent. One survey indicated that only 10% of the agencies charged fees, while another indicated nearly half did so — usually \$25 but one as high as \$145. Recent trends appear to indicate an increase in the providers charging at least nominal fees for their services.

In some places, lenders pay a fee to the HEC providers for each mortgage closed with one of their clients.

6.2.8. Effectiveness

The studies undertaken to date have not conclusively assessed the effectiveness of these programs in expanding homeownership while controlling the default risk. Efforts to evaluate their efficacy have been confounded by the diversity of their clientele and content, as well as their rapid growth and change. As the industry has matured, however, the variations are becoming somewhat less, and there are various studies now underway that might address this issue.

Based upon anecdotal findings, most HEC organizations questioned report default rates around 1%, which is about the average for all mortgages in the US. Some report even lower rates, while others higher. If the 1% figure is shown to be more or less true by empirical surveys, it represents an important indicator of the effectiveness of these programs.

6.3. Experience in Canada

Canada does not have a comparable HEC industry. In the absence of any similar financial and institutional support, various governmental, non-profit and for-profit organizations in the Canada have introduced a diverse range of initiatives, but no concerted effort has been made to expand that market by addressing the needs of the lower-income households being left out of homeownership.

The educations service available here have focused mostly on the dissemination of information, and mainly to the conventional homeowners market needing basic information about the home buying process and what is affordable.

The counselling programs in this country have similarly limited in number and scope. The nearest equivalents have been debt management counselling for families including, but not only, homeowners at risk of losing their homes, and one-on-one guidance associated with certain past government and non-profit housing programs directed at facilitating homeownership by lower-income households. Also, as indicated elsewhere in this report, a few organizations involved in rent banks and in IDA programs have started to incorporate financial counselling.

In the research design of *learn***\$ave** a specific analysis of the financial management training will be undertaken to identify the results and effectiveness of this support.

7. ASSESSMENT OF LICOS

This section examines the appropriateness of using Statistic Canada's Low-Income Cut-Offs (LICOs) as the measure of income-eligibility for IDA homeownership programs. The LICOs are the most widely used income measures in this country. In the absence of any standard approach for setting income thresholds in homeownership programs, various organizations, including the current IDA programs in Winnipeg and Calgary have adopted the LICOs for this purpose.

7.1. Background

The LICOs were prepared as a measure to assess income inequality in this country and particularly to monitor how inequality has changed over time.

Stats Canada has regularly published data on the proportion of Canadians falling below two income thresholds, the LICOs since 1967, and the Low Income Measure since 1991 (but with figures going back to 1980). Starting possibly in 2002, it also will provide data for a new measure, called the Market Basket Measure. All of these are defined later.

To fall below the LICO thresholds, a household (either individual or family) must spend 20% more of their income on food, shelter and clothing than is spent by the average household of their size living in a similarly sized community.

Statistics Canada (StatsCan) draws upon data from period surveys of family expenditures to determine what the average family spends on food, shelter and clothing. That dollar amount is converted into a percentage of the average family's income for that year. For the intervening years between the expenditure surveys, the percentage is increased to reflect that annual changes in the cost of living. The expenditures also are adjusted at various times to reflect changes in household expenditure patterns.

In setting the LICOs, StatsCan adds 20% to the percentage of average family's income spent on food, shelter and clothing. The reason for this 20% goes back to the initial formulation of the LICOs. The original expenditure data indicated that on average families spent 50% of their income on these necessities. StatsCan posited that families spending more than 70% of these necessities should be defined as being in financially difficult, or "straitened circumstances", because they would not have sufficient money to pay for all of the other essential expenditures (i.e., transport, personal care, household supplies, recreation, health and insurance.) Although the average expenditure on food, shelter and clothing has changed over the years, StatsCan has continued to apply the 20% in all subsequent calculations.

The LICOs vary according to the size of the household and the size of the community in which they live. The latter adjustment is meant to account for higher expenditures in the larger urban centres. As indicated, they are updated annually (see following table for the 2001 figures).

The LICOs are reported in relation to both pre and/or post-tax income. The pre-tax income refers to income before tax from salaries, investments and tax transfers like child tax benefits, Canada Pension plan payments and social assistance. The tax system in Canada has very significant impact on these figures when compared on a pre- and post-tax basis. The post-tax figures provide a better indication of the disposable income that is available to families.

According to StatsCan, the percentage of the family's total income spent on food, shelter and clothing has declined over time as the Canadian standard of living has improved. In 1959, the average expenditure was 50% of the total family income. According the last expenditure survey in 1992, the figure dropped to about 35%. As a result, the LICO thresholds have become based on a smaller percentage of the total household income.

7.2. LICOs as measures of poverty

The LICOs are the most widely accepted measure of poverty used in this country. Most social policy groups and many government bodies use the LICOs to define poverty thresholds.

Low Income Cut-Offs: 2002*

Household Size	Size of Urban Area by Population:						
	500,000 and over	100,000 to 499,999	30,000 to 99,999	less than 30,000			
1 person	\$22,609	\$19,392	\$19,258	\$15,625			
2 persons	\$28,261	\$24,240	\$24,072	\$19,530			
3 persons	\$35,148	\$30,148	\$29,938	\$24,290			
4 persons	\$42,546	\$36,493	\$36,240	\$29,402			
5 persons	\$47,560	\$40,794	\$40,510	\$32,868			
6 persons	\$52,573	\$45,095	\$44, 780	\$36,334			
7 persons+	\$57,586	\$49,396	\$49,050	\$39,799			

^{*} Based upon 2002 LICOs figures published by Statistics Canada.

This practice, nevertheless, has emerged without the sanction of StatsCan. The agency has always and continually stressed that it does not recognize LICO as a measure of poverty nor does it promote its use as such. StatsCan consistently refers to people with incomes below the LICOs, not as living in poverty, but as living in "straitened circumstances".

StatsCan conducted a broad review of the LICOs and various alternatives in 1989. The LICOs were found to have limitations, but no superior alternative could be identified as a replacement. So, StatsCan still continues to use the LICOs as its preferred measure of low income, and as a consequence, the measure is used for most policy initiatives addressing poverty.

In the absence of a clearly better replacement, there is an important reason for the continued use of the LICOs. Because they have been used over thirty years, a large body of social policy research and statistical data has been built upon them. Even if an alternative became available, the LICOs would still need to be used for some time as a reference point and for any trend analyses.

Unlike the US and some other countries, Canada has no official "poverty line".

The suitability of the LICOs as a measure of poverty, as well as various alternatives, has been debated for many years in this country. The following is only a brief summary of that debate.

7.2.1. Merits

The LICOs are also said to have the merit of being conceptually straightforward to construct and easy to apply. It is also said that they are easily understood, but that viewpoint is not universally shared.

To their credit, it should also be noted that the LICOs also appear to be a good indicator of the public's perception of poverty. Gallup have surveyed Canadians since 1976 on their view of income adequacy by asking this question: "What do you think is the least amount of money a family of four needs each week to get along in the community?" The Gallup results have been remarkably close to LICO figures for more than two decades.

There is another similar example. In Winnipeg in 1997, seven low-income respondents were asked to determine what would be included in the a basket of goods and services sufficient to provide and acceptable living level described as a "reasonable but not extravagant expectation of living costs." The results were just under the relevant LICO.

7.2.2. Limitations

Several criticisms have been raised about the LICO figures as a measure of poverty. One of the objections relates to the acknowledged arbitrariness of 20% figure. As indicated, the 20% figure was added to the initial average expenditure figure in order to top it up to 70%, which StatsCan believed to be the threshold at which families were living in "straitened circumstances". The 70% threshold no longer applies, but the 20% factor has been continued.

A more serious criticism in the context of this study is that the LICOs fail to reflect the high housing costs in certain major urban centres. Housing costs in these centres swallow a very high proportion of the income of low income households. Although the LICOs are adjusted according to the size of the community, those adjustments do not adequately reflect these differences in shelter costs. This aspect is examined more later in this section.

StatsCan's 1989 review of the LICOs identified a number of problems, some of which are identified above. Other related to certain methodological issues, and the infrequent occurrence of the expenditure surveys. The review also noted that the measure make no allowance for assets (like accumulated home equity) or for the value of in-kind transfers like subsidized rent.

There also have been objections to this "relative" approach to defining and measuring poverty, most notably and recently by Christopher Sarlo and the Fraser Institute. They argue that it measures not poverty, but income inequality, because the LICOs are based on a given family's expenditures relative to the average Canadian figure. Their alternative is to base the poverty line on the cost of basic necessities (see following material under 'Basic Needs Measure').

7.2.3. Relative vs. Absolute

Much of the on-going debate in Canada about the appropriateness of the various alternatives revolves around two opposing ways of thinking, one characterized as being absolute and the other as being relative.

Those in the absolute camp hold that poverty should be objectively defined, and the way to do this is to measure it against a subsistence standard of living. Their focus is on determining the income needed to provide a family with the basic "necessities" of life.

The relative camp holds that poverty should be seen essentially as an issue of social inequality. Their focus is on measuring the lack on income against some community norm (like average income) that reflects an appropriate standard of living or a reasonable share of wealth.

The various alternatives for measuring poverty income can be generally associated with one camp or the other. The LICOs, however, are an example of a hybrid measure that contains elements of both. To wit, the LICOs are based on a given family's expenditures for a set of necessities, but those expenditures are related back to the average figure.

Although the absolute approach is portrayed as being objective, determining what should be included in the basic necessities of life is not so easy. Therefore, this approach is ultimately based on what is a subjective choice.

One of the problems plaguing the debate is that the two approaches produce very different results. The relative measures typically higher poverty thresholds than the absolute. Furthermore, and even more important, the relative measures (based on the rise in the general standard of living or average income) will increase at a faster rate the absolute measures (based on the inflation in the cost of certain goods and service). In other words, the two approaches foster a different view about who should be considered to be poor, and the differences in those views would significantly widen over time.

All of this makes the choice of a measure for social policy programs at best a difficult task. Because poverty is a highly charged subject even without this problem, it is hard to bridge or reconcile these differences with any single measure.

7.2.4. Alternatives

Various alternatives to the LICOs have been developed, some in response to one or more of these objections.

a) Low Income Measure (LIM)

This measure was introduced by StatsCan in 1991 as alternative to the LICOs, following its 1989 review of the measure. While it has been used by analysts inside of StatsCan, it has not been much used by those outside.

LIM sets the low-income threshold at 50% of the median family income, when adjusted to take into account the needs of different family sizes. To determine the conversion factors for the family size adjustments, the first person in the household is counted as 1.0, the second person regardless of age is counted at 0.4, and after that each adult counts as 0.4 and each child as 0.3.

Similar yardsticks are used in other countries. Therefore, it provides a basis for international comparisons.

On this point, it is relevant to note that housing programs in the US all use some percentage of the median income as the basis for determining income eligibility. For example, homeownership programs typically use 80% of median income, and rental programs 50%.

b) Market Basket Measure (MBM)

This measure was released for consultation by HRDC in 1998, after being collaboratively developed by the federal, provincial and territorial governments. Its development is related to the National Child Benefit, the new federal program aimed at addressing child poverty. Various provincial governments wanted a measure that could used to evaluate its impact.

The measure is directed at identifying the minimum income required to meet the cost of a "basket" of defined "necessary" goods and services. The "basket" is defined not at an subsistence level, but at a level required for a basic, adequate standard of living. It includes food,

shelter clothing and transportation, and also incorporates a multiplier to cover other essentials. The costs of these needs are based on the actual costs in various communities across Canada.

The results are expressed in terms of the minimum disposable income needed to afford the basket of goods and services, and calculated for each province and for different sizes of community within each province. Disposable income is more restrictive than the after-tax income normally calculated by StatsCan. It excludes such expenses as support payments, work-related childcare costs and employee contributions to employment insurance.

The new measure was developed to meet various specifications, some of which were in response to the perceived shortcomings in the LICOs. The specifications included the following:

- It had to reflect a consensus view of the essential goods and services need to provide a minimum acceptable level of living.
- It had to be easy to understand (but this need not mean that it must be easy to calculate).
- It needed to be sensitive to geographical differences in the cost of living.
- It had to be capable of being adjusted annually to reflect the price differences and periodically to reflect changes in consumption patterns.

MBM remains controversial because it involves subjective judgements about what to include in the basket, and because it identifies a lower income for the poverty line when compared to the traditional pre-tax LICOs.

c) Basic Needs Measure

This measure is based on Christopher Sarlo's work on poverty, and has been promoted by the Fraser Institute. Sarlo, who has been outspoken critic of the LICOs, advocates a poverty line derived from the cost of a tightly defined set of basic necessities.

This measure is based upon the money needed for food, shelter, and clothing to keep the family alive. It is limited to the basic physical necessities, defined as the "items the absence of which is likely to compromise long term physical well being." It excludes as non-essential such items as books, toys, haircuts, dental services and school supplies. These are called "social comforts" or "social amenities".

Sarlo's poverty lines are very low compared to all other Canadian poverty lines. According to his calculations, in 1988 only one million Canadians lived in poverty. Some feel that this measure is a way of defining poverty out of existence.

7.3. LICOs as measures for homeownership

The question remains whether the LICOs, despite these other criticisms, can be used as an effective measure of income eligibility for any national homeownership like that envisioned by *Home***\$ave**.

The participants in any IDA homeownership program, it must be appreciated, will eventually need to meet the minimum household income requirements of the mortgage providers in order to buy a home. Those minimum income requirements will be largely determined by the standard lending criteria (specifically, the permitted gross debt service ratio), the current interest rates and the current house prices at the low end of the market.

Based upon all of these factors, the minimum annual household income needed currently to qualify for home purchase is about \$51,300 in Toronto, \$43,700 in Montreal and \$30,400 in the other three cities (see following table).

In comparison, the LICOs for a four-person household vary from \$35,500 to \$30,400 in the five cities, depending upon their size.

If the LICOs were used as the basis for determining eligibility in a national program, the relevant income thresholds would needed to be adapted city by city. As indicated by the above numbers, the thresholds would need to be set at something like 86% of LICO in Winnipeg, 100% in Saskatoon and Halifax, 123% in Montreal and 145% in Toronto.

This indicates that the LICOs would represent a poor measure of income eligibility for a national homeownership program. Although the LICOs are adjusted to reflect the higher cost in larger centres, these adjustments do not fully reflect the higher shelter costs that occur in the more expensive cities like Toronto, Montreal and presumably other places as well.

Although the LICOs appears to be close to the relevant income threshold in the three cities with lowest house prices, it should be noted this would change if the interest rates rose back to their historic norm, and if the particular households were already carrying a significant debt load.

In any case, any measure used principally to define poverty thresholds is not likely to be relevant to a program promoting homeownership. Households at or near poverty are not likely candidates for homeownership, unless the program is capable of providing deep subsidies.

7.3.1. Comparison of the LICOs to Mortgage Requirements

	Winni- peg	Sas- katoon	Tor- onto	Mon- treal	Hali- fax
Min House Price:	\$70,000	70,000	140,000	110,000	70,000
Mortgage Loan:	\$66,500	66,500	133,000	104,500	66,500
Min Household Income for Mortgage Loan:	\$30,400	30,400	51,300	43,700	30,400
2001 LICO (4 pers):	\$35,500	30,400	35,500	35,500	30,400
Comparison with Min Income for Loan:	86%	100%	145%	123%	100%

Assumptions:

Min house price - based upon the lowest-priced two-bedroom houses on market (MLS, March 2002), but adjusted upwards on advice of stakeholders in each city to include houses only in good repair.

Mortgage Loan - based upon 95% of minimum house price.

Min Household Income - based upon a mortgage interest of 71/4% for a 25-year amortization period and a gross debt service ratio of 32%.

APPENDIX 1: LIST OF PARTICIPANTS AT THE STAKEHOLDERS' MEETINGS

The following stakeholders attended the meetings in the five cities.

Saskatoon - 16 May 2002

Russell Mawby Housing Facilitator City of Saskatoon 306-975-7666

russell.mawby@city.saskatoon.sk.ca

Brenda Wallace Executive Director Saskatoon Housing Initiatives Partnership 306-934-1711 brenda.wallace@sk.sympatico.ca

Norm Klatt Manager, Deposit & Lending Services Co-operative Trust Co of Canada 306-956-1936 nklatt@co-operativetrust.ca

Keith Hanson Director Affordable New Home Development Foundation 306-665-2525 khanson.sr@sk.sympatico.ca

Sheri Benson Director of Community Services United Way of Saskatoon 306-975-7705 sbenson@saskatoon.unitedway.ca

Jim Clarke
Business Development Co-ordinator
Quint Development Corp
306-978-0163
businessdev_quint@sasktel.net

Rob Watts Community Development Manager Saskatoon Region Saskatchewan Social Services 306-933-6067 rwatts@ss.gov.sk.ca

Martin Chicilo Community Development Officer Saskatoon Credit Union 306-934-4052 martin.chicilo@saskatoon.cu.sk.ca

Ruth Pepin

Executive Director
The Saskatoon Foundation
306-665-1766
ruth.peppin@thesaskatoonfoundation.ca

Roger Parent Development Officer Metis Nation of Saskatchewan 306-244-7441 rparent.mns@sasktel.net

Bill Joyner
Partnership Consultant
CMHC Centre for Public-Private Partnerships - Calgary
403-515-3011
bjoyner@cmhc-schl.gc.ca

Jim Wasilenko General Manager Saskatoon Housing Authority 204-668-2703 jwasilenko@sk.sympatico.ca

Lyn Gray Senior Development Officer HRDC 306-975-4499 lyn.gray@hrdc-drhc.gc.ca

Rob Harasymchuk Aboriginal Human Resources Development Council of Canada 866-711-5091 rob.harasymchuk@ahrdcc.gc.ca

invited and confirmed but not attending

Kelly Lindsay President Aboriginal Human Resources Development Council of Canada 866-711-5091 kelly.lindsay@ahrdcc.gc.ca

Winnipeg - 16 May 2002

Peter Squire
Director of Housing Opportunities Program (HOP)

Winnipeg Real Estate Board 204-786-8854 peter.squire@wreb.ca

Tom Carter Professor Institute of Urban Studies University of Winnipeg 204-982-1148 t.carter@UWinnipeg.ca

Sheri Blake Assistant Professor Dept of City Planning University of Manitoba 204-474-6426 blakes@cc.umanitoba.ca

Russ Rothney Manager of Community Economic Development Assiniboine Credit Union 204-958-8756 rothneyr@assiniboine.mb.ca

Linda Ring RRAP Co-ordinator City of Winnipeg 204-940-3080 liring@gov.mb.ca

Shannon Watson
Executive Director
Spence Neighbourhood Association
204-783-5000
sdpark@mb.sympatico.ca

Brian Grant Housing Co-ordinator West Broadway Development Corporation 204-975-1656 bgrant@westbroadway.mb.ca

Roy Kirby Provincial Director Winnipeg Housing and Homeless Initiative 204-940-3072 rkirby@gov.mb.ca Dennis Cleve CMHC Representative in Manitoba Winnipeg Housing and Homeless Initiative 204-983-8503 dcleve@cmhc-schl.gc.ca

Stephano Grande General Manager Winnipeg Housing Rehabilitation Corporation 204-949-2886 sgrande@shaw.ca

Bern Bileski Director of Community Investments United Way of Winnipeg 204-477-5360 bbileski@unitedwaywinnipeg.mb.ca

Wayne Helgason Executive Director Social Planning Council Winnipeg 204-943-2561 x224 wayneh@spcw.mb.ca

Susan Deane Executive Assistant to the Minister Manitoba Housing and Family Services 204-945-8208 sdeane@leg.gov.mb.ca

Rev. Harry Lehotsky New Life Ministries 204-775-4929 lehotsky@escape.ca

Brian Williamson Communications Coordinator - HRDC Winnipeg Housing and Homeless Initiative 204-984-0592 brian.williamson@hrdc.drhc.gc.ca

invited and confirmed but not attending

Mary Williams General Manager North End Housing Project 204-953-1895

Ken McMurdo General Manager Me-dian Credit Union 204-925-9111 kmurdo@mediancu.mb.ca

Myles Livingstone Manager, HRDC Housing and Homeless Initiative 204-983-7313 myles.livingstone@hrdc-drhc.gc.ca

Toronto - 31 May 2002

Fred Hayes
Program Coordinator, Community and Cultural Markets
RBC Royal Bank
416-974-3050
fred.hayes@rbc.com

Greg Suttor
Policy Development Officer
City of Toronto, Social Development and Administration
416-397-5810
gsuttor@city.toronto.on.ca

Linda Belanger Director of Business Development GE Capital Mortgage 905-858-5475 linda.belanger@gecapital.com

Mark Guslits
Special Advisor: Let's Build Program
City of Toronto, Shelter, Housing and Support
416-397-4164
mguslits@city.toronto.on.ca

Maureen Adams
Vice President of Allocations and Community Service
United Way of Greater Toronto
416-777-2001
kazoulay@uwgt.org

Ray Hull

Consumer Lender Team Leader Metro Credit Union 416-252-5625 x7694 rhull@metrocu.com

Steve Pomeroy President Focus Consulting 613-722-0110 spomeroy@rogers.com

Angus Palmer General Manager Wigwamen Inc. 416-481-4451 apalmer@wigwamen.com

Antoine Pomerleau Regional Consultant, Homelessness Partnerships HRDC 416-954-7210 antoine.pomerleau@hrdc-drhc.gc.ca

Diane Kinney
Ontario Homelessness Directorate
HRDC
416-954-7128
diane.kinney@hrdc-drhc.gc.ca

Robert Freeman
Community Development Officer - Social Housing
Homeless and Housing Initiatives
905-453-1300 x2523
freemanr@region.peel.on.ca

invited and confirmed but not attending

John Metson Manager, CRC Self Help 416-591-6389 johnm@tcrc.on.ca

Ivy France
Housing Policy Supervisor, Policy Section, Housing and Property Services Department
Region of Peel
905-453-1300 x2670
ivy.france@region.peel.on.ca

Tim Welch Housing Programs Administrator Region of Waterloo 519-575-4005 wetim@region.waterloo.on.ca

Bill Beatty
Manager, Assisted Housing Centre, Ontario Region
CMHC
416-218-3308
wbeatty@cmhc-schl.gc.ca

John Fogolin President Tornat Construction 905-738-1660 jfogolin@hotmail.com

John Jagt
Director of Emergency Services
City of Toronto, Shelter, Housing and Support
416-392-5358
jjagt@city.toronto.on.ca

Halifax - 7 June 2002

Jim Graham
Director of Housing Services
Province of Nova Scotia
902-424-3224
grahamjd@gov.ns.ca

Carla Staresina
Principal
CMHC Centre for Public-Private Partnerships
902-426-8430
cstaresi@cmhc-schl.gc.ca

Wayne Paterson Manager, Lending Services League Savings and Mortgage 902-454-3135 wpaterson@lsm.ca

Patricia Richards Community Co-ordinator Community Action for Homelessness 902-420-2186 patriciar@hfx.eastlink.ca

Rick Parker Manager, Consulting and Support Credit Union Central of Nova Scotia 902-454-3125 rparker@cucns.ca

Barb Nehiley
Special Projects Advisor
Strategic Information, Program Design and Project Management
Halifax Regional Municipality
902-490-4612
nehileb@region.halifax.ns.ca

Natasha Jackson Manager, Communications Metropolitan Region Housing Authority 902-420-2167 jacksonl@nsrha.gov.ns.ca

Tom Levesque
Manager of Assisted Housing
CMHC
902-426-1811
tlevesqu@cmhc-schl.gc.ca

Douglas Sparks
Development Co-ordinator
Lucasville and Upper Hammonds Development Association
902-835-1441
dsparks@luhp.ca

invited and confirmed but not attending

Dean Durnford Principal for Single Underwriting Business Development for Atlantic Canada - CMHC 902-426-8435 ddurnfor@cmhc-schl.gc.ca

Nancy Wooden General Manager Harbour City Homes Halifax Non-Profit 902-490-6277 Lynn Drinovz Cooperative Housing Federation of NS 902-455-0470 chfns@chebucto.ns.ca

Sidney Peters
Executive Director
Ki'knu Housing
902-893-9804
sidneypeters@tru.eastlink.ca

Montreal - 19 June 2002

Allan Gaudreault Consultant - Habitation et développement communautaire 514-278-1043 allan.gaudreault@videotron.ca

Robert Cohen Consultant - Habitation et développement communautaire 514-484-1405 chaltfim@netrover.com

Jean-Jacques Bohémier Conseiller politique - Habitation Cabinet du maire et du comité exécutif Ville de Montréal 514-872-1233 jbohemier@ville.montreal.qc.ca

Martin Wexler
Chef de division
Expertise et du soutien au dévelopment résidentiel
Direction de l'habitation
Ville de Montréal
514-872-5477/514-872-8237
mwexler@ville.montreal.qc.ca

James McGregor
Deputy Director of Housing
Ville de Montréal
514-872-9251
jmcgregor@ville.montreal.qc.ca

Jeanne Wolfe Professor, School of Architecture McGill University 514-398-4077 jeanne.wolfe@mcgill.ca

Giséle Rucker Assistance Executive Directive The Samuel and Saidye Bronfman Family Foundation 514-878-5274 giselerucker@claridgeinc.com

Marie Murphy Conseillère principale, Partenariat public-privé Centre d'aide au logement Société canadienne d'hypothèques et de logement 514-283-2203 mmurphy@cmhc-schl.gc.ca

Catherine O'Neill Project Co-ordinator Groupe CDH 514-849-7800 catherine@groupecdh.com

Marie-Helene Legault
L'ACEF del'Est (l'Association coopérative d'économie familiale)
514-257-6622
acefest@consommateur.qc.ca

André Noël General Manager Fédération des coopératives d'habitation de I'île de Montréal 514-843-6929 #235 anoel@fechim.qc.ca

François Thérien Direction de la santé publique de Montréal 514-528-2400 x3414 ftherien@santepub-mtl.qc.ca

Colette Basque Societe d'habitation du Quebec (SHQ) 418-644-8872 colette.basque@shq.gouv.qc.ca Renée Lachapelle Families en tête 514-932-8111 rlachapelle@fet.qc.ca

invited and confirmed but not attending

Diane Paquin
Conseillère financement aux enterprises
Caisse d'économie Desjardins des Travailleuses et
Travailleurs
514-598-2479/598-2122
diane.paquin@desjardins.com

Odette Viens Housing Advisor Centraide du Grand du Montréal 514-288-1261 vienso@centraide-mtl.org

Marc Godbout Consultant - Housing Finance Infuni inc 450-674-0662 infuni@aei.ca

APPENDIX 2: REFERENCES

Asset-Based Policies

Boshara, Ray (editor): Building Assets - A Report on the Asset-Development and IDA Field: Corporation for Enterprise Development: December 2001. Available at:

http://www.idanetwork.org/index.php?section=resources&page=publications.html

Boshara, Ray; Poverty Is More Than a Matter of Income; New York Times: 29 September 2002. Article available at: http://www.newamerica.net/index.cfm?pg=article&pubID=981

Edwards, Laura: Equity stakes, fair stakes; Institute for Public Policy Research, London, England: September 2001. Focus group findings - IPPR, from group research with Council & Housing Association Tenants.

Available at:

http://www.ippr.org.uk/research/files/team27/project64/equitystakes110901.pdf

Friedman, Robert E. and Sherraden, Michael (2001): Asset-Based Policy in the United States, in Sue Regan and Will Paxton (editors); Asset-Based Welfare: International experiences. London: The Institute for Public Policy Research.

Available at:

http://gwbweb.wustl.edu/csd/Areas_Work/Asset_building/ADD/ADD_publications.htm

Grant Insights: Assessing the Case for Asset-Based Social Policy in Canada; Submission to Social and Enterprise Development Innovations: August 1998.

Available at: http://www.grantinsights.com/Advocacy.html

Email: michael@grantinsights.com

Gosse, Barbara A., Nares, Peter and Robson-Haddow, Jennifer: *The Case for an Asset-based Approach to Social Policy in Canada, Asset-Based Welfare: International Experiences, London, England: The Institute for Public Policy Research, London England, 2001.*

Lam, Sophie et al. Individual Development Accounts in Canada; A Return on Investment Demonstration Model; prepared for Social and Enterprise Development Innovations: September 1998. Available at: http://www.sedi.org/DataRegV2-unified/sedi-Reports/IDA%20ROI.PDF

Nares, Peter: Building Assets and Opportunity: Towards a New Vision for Social Policy in Canada; Social and Enterprise Development Innovations: April 2002.

Paxton, Will and Regan, Sue: Policy Options for the Progressive Encouragement of Assets, in Ownership for All, published by Gavin Kelly and Rachel Lissauer, Institute for Public Policy Research, London, England, February 2000. Institute for Public Policy Research, London England.

Available at: http://www.policylibrary.com/Essays/IPPRassets/assetsippr3.htm

Key Links: http://www.ippr.org.uk/search/index.php

Paxton, Will: Asset-based Policy in the United Kingdom, Institute for Public Policy Research; London,

England: 2002.

Paxton, Will and Regan, Sue: *Progressing Asset-Based Policies in the UK*, paper prepared for International Assets Seminar: Institute for Public Policy Research, London, England: 2001. Available at: http://gwbweb.wustl.edu/csd/News/Blair.pdf

Schreiner, Mark et al: Savings and Asset Accumulation in Individual Development Accounts; Center for Social Development, February 2001.

Available at:

http://gwbweb.wustl.edu/csd/Publications/2001/ADDreport2001/title&contents.pdf

IDA Programs

Center for Social Development: *Individual Development Accounts*: 1999.

Available at: http://gwbweb.wustl.edu/csd/Areas Work/Asset building/IDAs/index.htm

Center for Social Development: *IDA Policy in the State*; September 2001. Available at: http://gwbweb.wustl.edu/csd/statepolicy/StateIDAtable.pdf

Corporation for Enterprise Development: 2002 Federal IDA Briefing Book: How IDAs Affect Eligibility for Federal Programs: 2002.

Available at: http://www.cfed.org/individual assets/2002 Federal IDA Briefing Book.pdf

MCC Employment Development: Asset Development.

Available at:

http://www.mcca-ed.org/services/programmes/fl/asset_development.html

Mills, Gregory et al: Evaluation of Asset Accumulation Initiatives, Final Report, Abt Associates: prepared for the US Dept of Agriculture: 29 February 2000. Available at: http://www.fns.usda.gov/oane/MENU/Published/FSP/FILES/Program%20Design/AssetAcc.pdf

Saul, Richard: Assets for Independence Demonstration: 2001. Available at: www.idanetwork.org/archives/idanetwork/aug2001/att-0007/04-2001 2ND ANN SUMMARY.DOC.

Social and Enterprise Development Innovations (SEDI): *learn*\$ave *National Demonstration of IDAs* for Learning.

Available at: mww.sedi.org:80/html/prog/save/prog-save-learn.asp

Federal Homeownership Programs

Canada Customs and Revenue Agency:

Available at: http://www.ccra-adrc.gc.ca/tax/registered/resp-e.html

Canada Mortgage and Housing Corporation: Home Buyers' Plan Helps Turn Homeownership Dream into Reality; Housing Facts; Vol. 5, No 4: 10 April 2000.

Available at: http://www.cmhc-

schl.gc.ca/en/homadoin/maintrst/foan/loader.cfm?url=/commonspot/security/getfile.cfm&PageID=46228

Canada Mortgage and Housing Corporation: Home Buyers' Plan Turns Home Ownership Dream Into Reality; CMHC New Release: 29 March 2001.

Available at: http://www.cmhc-schl.gc.ca/en/News/nere/2001/2001-03-29-1400.cfm

Canada Mortgage and Housing Corporation: Mortgage Insurance: May 2002.

Frenken, Hubert: *The RRSP Home Buyers' Plan*; PERSPECTIVES; Statistics Canada: Summer 1998.

Available at:

http://www.statcan.ca/english/indepth/75-001/archive/1998/pear1998010002s2a05.pdf

Independent Mortgage Brokers Association of Ontario: *Homeowner Mortgage Loan Insurance*. Available at: http://www.imba.ca

Toronto Real Estate Board: RRSP Home Buyers' Plan: May 1998. For General information on how to use RRSP's toward the purchase of a home. Available at:

http://www.mls.ca/boards/treb/gov_prog/rrsp.htm

Toronto Real Estate Board: Toronto Real Estate Board's Legislative Priorities; First Home Loan Insurance Program; Hot Issues: May 1992. Available at:

http://www.homestore.ca/tabs/homesforsale/toronto/news/pac/ISS0399/fed_iss/hom_loan_.htm

Provincial and Territorial Housing Agreements

Canada Housing and Mortgage Corporation: Canada and Northwest Territories Sign Affordable Housing Agreement: 5 February 2002.

Available at: http://www.cmhc.ca/en/News/nere/2002/2002-02-05-0000.cfm

Canada Mortgage and Mortgage Corporation: Canada and Nunavut Sign Affordable Housing Agreement: February 2002.

Available at: http://www.cmhc.ca/en/News/nere/2002/2002-02-06-1600.cfm

Canada Mortgage and Housing Corporation: Canada and Ontario Sign Affordable Housing Agreement: May 2002.

Available at: http://www.cmhc-schl.gc.ca/en/News/nere/2002/2002-05-30-1530.cfm

Canada Mortgage and Housing Corporation: Canada and Alberta Sign Affordable Housing Agreement: 24 June 2002.

Available at: http://www.cmhc-schl.gc.ca/en/News/nere/2002/2002-06-24-0000.cfm

Canada Mortgage and Housing Corporation: Canada and Saskatchewan Sign Affordable Housing Agreement: 28 June 2002.

Available at: http://www.cmhc-schl.gc.ca/en/News/nere/2002/2002-06-28-1530.cfm

Canada Mortgage and Housing Corporation: Canada and Nova Scotia Sign Affordable Housing Agreement: 13 September 2002.

Available at: http://www.cmhc-schl.gc.ca/en/News/nere/2002/2002-09-13-0930.cfm

Canada Mortgage and Housing Corporation: Canada and Manitoba Sign Affordable Housing Agreement: 20 September 2002.

Available at: http://www.cmhc-schl.gc.ca/en/News/nere/2002/2002-09-20-0330.cfm

Canada Mortgage and Housing Corporation: Canada and Yukon Sign Affordable Housing Agreement: 3 October 2002.

Available at: http://www.cmhc-schl.gc.ca/en/News/nere/2002/2002-10-03.cfm

Provincial and Municipal Homeownership Programs

Affordable Home Ownership Policy, City of Surrey: January 2002. Available at: http://www.gvrd.bc.ca/services/housing/Surrey Home Ownership Program/program policy.pdf

Home Ownership Program: Department of Family and Community Services, New Brunswick: 2002.

Available at: www.gnb.ca/0017/housing/ownership-e.asp

Home Ownership Assistance Program, City of Surrey: January 2002. Available at:

http://www.gvrd.bc.ca/services/housing/Surrey Home Ownership Program/program brochure.pdf

Home Ownership Assistance Program Guide: City of Surrey: 2002. Available at: http://www.gvrd.bc.ca/services/housing/Surrey Home Ownership Program/program guid e.pdf

Ontario Home Ownership Savings Plan: Information Bulletin (Dist 6305), Ontario Ministry of Finance:

April 2001.

Available at: http://www.trd.fin.gov.on.ca/userfiles/page attachments/6305 OHOSP.pdf

Ontario Home Ownership Savings Plan: Ottawa Real Estate Board: 2002.

Available at: http://orebweb1.oreb.ca/frm info.html

Neighbourhood Home Ownership Plan:

Available at: http://www.gov.sk.ca/socserv/housing/programs/p-NHOP.html

Wexler, Martin E.: Home Ownership Strategies: An Integral Part of Neighbourhood Revitalization: Institute of Urban Studies: May 2002.

Local IDA Programs in the US

Housing Counselling & The IDA Program: The Affordable Housing Coalition of Asheville & Buncombe Count.

Available at: http://www.ahcabc.org/news/news.html

Individual Development Account Program: Columbus Housing Partnership, Montgomery Housing Partnership.

Low-Income Families Can Double Their Savings: DuPage County IDA Program.

Article currently available at: http://www.vdcom.com/articles/publish/article 138.shtml

State-by-state listings of IDA programs: Available at: www.idanetwork.org/index.php?section=state&page=state.php
http://www.idanetwork.org/index.php?section=state&page=state.php

Homeownership Education

Canada Mortgage and Housing Corporation: *Student Text - Home Ownership Education Training Program*: January 2002.

Canada Mortgage and Housing Corporation: Hirshhorn Consulting Inc.: Homeownership Education and Counselling: An Examination of US Experience and Its Relevance for Canada: June 2000. Available at: http://www.cmhc-schl.gc.ca/publications/en/rh-pr/socio/socio068.pdf

Rent Banks / Housing Trust Funds

Centre for Research and Education in Human Services: A Rent Bank for Waterloo Region: A Feasibility Study: 2001.

Drdla, Richard and Associates: Housing Trust Funds: Their Nature, Applicability and Potential in Canada; CMHC, 2000.

Hoy, Shirley: *Housing in the City,* City of Toronto: December 1999. Available at: http://www.city.toronto.on.ca/housing/house/pub/

LaPointe, Linda: Canada Mortgage and Housing Corporation: City of Toronto Rent Bank Pilot Project Evaluation: Research Highlights; Socio-Economic Series, Issue 57: 2001.

Social Services, City of Toronto: *Shelter Fund Enhancement Policy:* September 2000. Available at www.city.toronto.on.ca/socialservices/Policy/Shelterfund.htm. Available at: http://dsp-psd.communication.gc.ca/Collection/NH18-23-57E.pdf

Statistical Data Report: Housing Help Rent Bank Project, City of Toronto: 30 January 2002.

Wingard, Jeff: An Analysis and Review of Hamilton's Housing Emergency Loan Program. Social Planning & Research Council of Hamilton-Wentworth: September 2001.

Affordability Calculations

Canada Mortgage and Housing Corporation: Homebuying Step by Step - A Consumer Guide and Workbook:

Available at:

http://www.cmhcschl.gc.ca/en/bureho/buho/loader.cfm?url=/commonspot/security/getfile.cfm&PageID=38249

Starter Home Property Taxes and Heating Costs: September 1998.

Available at: http://www.cmhc-schl.gc.ca/en/imquaf/fime/afhofiso/afhofiso 011.cfm

The Home You Can Afford: Available at: www.cmhc.ca/en/moinin/moinbuho/brhoowwire/brhoowre 003.cfm

Poverty Measures

Hale, Alison: Poverty and Low Income Measurement in Canada, Recent Analyses and Future Directions,

Statistics Canada: November 1999.

Available at: http://www.ibge.gov.br/poverty/pdf/alison_hale.pdf

Jackson, Andrew: Low Income Trends in the 1990s; Canadian Council on Social Development: January 2001. Available at: www.ccsd.ca/pubs/2000/lit/

Paquet, Bernard: Low Income Cutoffs from 1991 to 2000 and Low Income Measures from 1990 to 199, Income Statistics Division, Statistics Canada: November 2001.

Available at:

http://www.statcan.ca/english/research/75F0002MIE/75F0002MIE2001007.pdf

Sarlo, Christopher A.: Measuring Poverty in Canada; Critical Issues Bulletin; Fraser Institute: 2001. Available at: http://oldfraser.lexi.net/publications/critical issues/2001/poverty/

Sherraden, Michael J.: Statistics Canada's Experience with Low Income Cut Offs, Household Surveys Division, Statistics Canada: 1996. Available at: <a href="http://www.statcan.ca:8081/english/clf/query.html?qt=low+income+cut+offs&ht=0&qp=&qs=&qc=0&pw=100%25&la=en&qm=1&st=1&oq=&rq=0&si=0&rf=0&col=alle

Shillington, Peter: A Poor Measure of Poverty.

Available at http://www.straightgoods.com/item39.asp

Other General Sources of Information

Assets - A Quarterly Update for Innovators published by Center for Enterprise Development.

Center for Enterprise Development at www.cfed.org

Ida Network at www.idanetwork.org

Visit our home page at www.cmhc.ca